%% @texfile{
%%     filename="KdVShockSoliton.tex",
%%     version="1.0",
%%     date="Sept-2021",
%%     cdate="20210501",
%%     filetype="LaTeX2e",
%%     pics="KdVShockSoliton1",
%%     journal="J. Math. Anal. Appl. 514, 126251 (2022)",
%%     doi="10.1016/j.jmaa.2022.126251",
%%     copyright="Copyright (C) I. Egorova, J.Michor and G. Teschl"
%%     }

\documentclass{amsart}
\usepackage{hyperref}

\usepackage{graphicx}
\usepackage{enumerate}
\usepackage{enumitem}
%\usepackage{showkeys}
%\usepackage{refcheck}
%\usepackage{todonotes}
\setitemize{leftmargin=7mm}
\setenumerate{leftmargin=7mm}
\unitlength1cm
%%%%%%%%%
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{arXiv:#1}}
\newcommand*{\mailto}[1]{\href{mailto:#1}{\nolinkurl{#1}}}

%%%%%%%%%THEOREMS%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newtheorem{theorem}{Theorem}[section]
\newtheorem{RHproblem}{RH problem}
\newtheorem*{modelRHP}{Model RH problem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}

\theoremstyle{definition}
\newtheorem{remark}[theorem]{Remark}

%%%%%%%%%%%%%%FONTS%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\R}{\mathbb{R}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\T}{\mathbb{T}}
\newcommand{\M}{\mathbb{M}}
\newcommand{\D}{\mathbb{D}}
\newcommand{\X}{\mathbb{X}}
\newcommand{\F}{\mathcal{F}}
\newcommand{\RR}{\mathcal{R}}
%%%%%%%%%%%%%%%%%%ABBRS%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\nn}{\nonumber}
\newcommand{\be}{\begin{equation}}
\newcommand{\ee}{\end{equation}}
\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\eea}{\end{eqnarray}}
\newcommand{\ul}{\underline}
\newcommand{\ol}{\overline}
\newcommand{\pa}{\partial}
\newcommand{\ti}{\tilde}
\newcommand{\wti}{\widetilde}
\newcommand{\wha}{\widehat}
\newcommand{\norm}[1]{\lVert#1 \rVert}
\newcommand{\abs}[1]{\lvert#1 \rvert}


\newcommand{\spr}[2]{\langle #1 , #2 \rangle}
\newcommand{\id}{\mathbb{I}}
\newcommand{\I}{\mathrm{i}}
\newcommand{\E}{\mathrm{e}}
\newcommand{\ind}{\mathop{\mathrm{ind}}}
\newcommand{\sgn}{\mathop{\mathrm{sgn}}}
\newcommand{\clos}{\mathop{\mathrm{clos}}}
\newcommand{\re}{\mathop{\mathrm{Re}}}
\newcommand{\im}{\mathop{\mathrm{Im}}}
\DeclareMathOperator{\Mod}{mod}
\newcommand{\sech}{\mathop{\mathrm{sech}}}
%\newcommand{\sinh}{\mathop{\mathrm{sinh}}}
\DeclareMathOperator{\res}{Res}
%\newcommand{\db}{\mathfrak{D}}
\newcommand{\noprint}[1]{}

%%%%DLMF%%%%
\makeatletter
\newcommand{\dlmf}[1]{%
\cite[%
  \def\nextitem{\def\nextitem{, }}%
  \@for \el:=#1\do{\nextitem\href{http://dlmf.nist.gov/\el}{(\el)}}%
]{dlmf}%
}
\makeatother

%%%%%%%%%%%%%%%GREEK%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\eps}{\varepsilon}
\newcommand{\vphi}{\varphi}
\newcommand{\si}{\sigma}
\newcommand{\la}{\lambda}
\newcommand{\ga}{\gamma}
\newcommand{\om}{\omega}


%%%%%%%%%%%%%%%%%%%%%%%%NUMBERING%%%%%%%%%%%%%%%%%%%%

\renewcommand{\labelenumi}{(\roman{enumi})}
\numberwithin{equation}{section}


%%%%%%
\newcommand{\sigI}{\begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}}
\newcommand{\ssigI}{\big(\begin{smallmatrix} 0 & 1 \\ 1 & 0\end{smallmatrix}\big)}
\newcommand{\rI}{\begin{pmatrix}  1 & 1 \end{pmatrix}}
\newcommand{\rN}{\begin{pmatrix}  0 & 0 \end{pmatrix}}
\DeclareMathOperator{\Ai}{Ai}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{XXX}

\bibitem{BET} A. Boutet de Monvel, I. Egorova, and G. Teschl,
 {\em Inverse scattering theory for one-dimensional Schr\"odinger
operators with steplike finite-gap potentials},
J. Analyse Math. {\bf 106} (2008), 271--316.
\url{https://doi.org/10.1007/s11854-008-0050-4}

\bibitem{BF} V.S. Buslaev and V.N. Fomin, {\em An inverse scattering problem for the
one-dimensional Schr\"odinger equation on the entire axis},
Vestnik Leningrad. Univ. {\bf 17} (1962), 56--64.

\bibitem{BT} D. Bilman and T. Trogdon, {\em On numerical inverse scattering for the Korteweg--de Vries equation with discontinuous 
step-like data}, Nonlinearity {\bf 33:5} (2020), 2211--2269.
\url{https://doi.org/10.1088/1361-6544/ab6c37}

\bibitem{CK} A. Cohen and T. Kappeler,
{\em Scattering and inverse scattering for steplike potentials in the Schr\"odinger equation},
Indiana Univ. Math. J. {\bf 34} (1985), 127--180.

\bibitem{DT}
 P. Deift and E. Trubowitz,
 {\em Inverse scattering on the line},
 Comm. Pure Appl. Math. {\bf 32} (1979), 121--251.

\bibitem{DZ} P. Deift and X. Zhou, {\em A steepest descent method for oscillatory
Riemann--Hilbert problems}, Ann. of Math. (2) {\bf 137} (1993), 295--368.
\url{https://doi.org/10.2307/2946540}

\bibitem{EGKT} I. Egorova, Z. Gladka, V. Kotlyarov, and G. Teschl,
{\em Long-time asymptotics for the Korteweg-de Vries equation with steplike initial data},
Nonlinearity {\bf 26} (2013), 1839--1864. \url{https://doi.org/10.1088/0951-7715/26/7/1839}

\bibitem{EGLT} I. Egorova, Z. Gladka, T.-L. Lange, and G. Teschl, {\em Inverse scattering theory for Schr\"{o}dinger operators with steplike potentials}, Zh. Mat. Fiz. Anal. Geom. {\bf 11} (2015), 123--158.

\bibitem{EGT} I. Egorova, K. Grunert, and G. Teschl,
{\em On the Cauchy problem for the Korteweg--de Vries equation with steplike finite-gap initial data I. Schwartz-type perturbations}, Nonlinearity {\bf 22} (2009), 1431--1457.
\url{https://doi.org/10.1088/0951-7715/22/6/009}

\bibitem{ET3} I. Egorova and G. Teschl,
{\em On the Cauchy problem for the Korteweg--de Vries equation with steplike finite-gap initial data II. Perturbations with finite moments}, J. d'Analyse Math. {\bf 115} (2011), 71--101.
\url{https://doi.org/10.1007/s11854-011-0024-9}

\bibitem{gp1} A.V. Gurevich and L.P. Pitaevskii, {\em Decay of initial discontinuity in the Korteweg--de Vries equation}, JETP Letters {\bf 17:5} (1973), 193--195.
\bibitem{gp2} A.V. Gurevich and L.P. Pitaevskii, {\em Nonstationary structure of a collisionless shock wave}, Soviet Phys. JETP {\bf 38:2} (1974), 291--297.

\bibitem{GR} S. Grudsky and A. Rybkin,
{\em On classical solutions of the KdV equation}, Proc. Lond. Math. Soc. (3) {\bf 121:2} (2020), 354--371.
\url{https://doi.org/10.1112/plms.12326}

\bibitem{GT} K. Grunert and G. Teschl,
{\em Long-time asymptotics for the Korteweg--de Vries equation via nonlinear steepest descent},
Math. Phys. Anal. Geom. {\bf 12} (2009), 287--324. \url{https://doi.org/10.1007/s11040-009-9062-2}

\bibitem{its1} A.R. Its, {\em Asymptotic behavior of the solutions to the nonlinear Schr\"odinger equation,
and isomonodromic deformations of systems of linear differential equations}, Soviet. Math. Dokl. {\bf 24:3} (1981), 452--456.

\bibitem{Kh} E.Ya. Khruslov,
{\em Asymptotics of the Cauchy problem solution to the KdV equation with step-like initial data},
Matem.sborn. {\bf 99} (1976), 261--281.

\bibitem{Lax} P.D. Lax, {\em Integrals of nonlinear equations of evolution and solitary waves}, Comm. Pure and Appl. Math. {\bf 7} (1968), 159-193.

\bibitem{Lau1} T. Laurens, {\em Global well-posedness for $H^{-1}(\R)$  perturbations of KdV with exotic spatial asymptotics}, \arxiv{2104.11346}, 2021.

\bibitem{Lau2} T. Laurens, {\em KdV on an incoming tide}, \arxiv{2104.11748}, 2021.

\bibitem{ma} S.V. Manakov, {\em Nonlinear Frauenhofer diffraction}, Sov. Phys. JETP {\bf 38:4} (1974), 693--696.

\bibitem{Len18} J. Lenells, {\em Matrix Riemann--Hilbert problems with jumps across Carleson contours}, 
Monatsh. Math. {\bf 186} (2018), 111--152. \\
\url{https://doi.org/10.1007/s00605-017-1019-0}

\bibitem{Lenells} J. Lenells, {\em The nonlinear steepest descent method for Riemann-Hilbert problems of low regularity},
Indiana Univ. Math. J. {\bf 66:4} (2017), 1287--1332.

\bibitem{M} V.A. Marchenko,
{\em Sturm--Liouville Operators and Applications} (Russian),
Naukova Dumka, Kiev, 1977;
Tanslated from Russian: Birkh\"auser, Basel, 1986;
rev. ed., Amer. Math. Soc., Providence, 2011.

\bibitem{Olver} F.W.J. Olver, {\em  NIST handbook of mathematical functions}, Cambridge University Press, 2010.

\bibitem{Zakh} V.E. Zakharov, S.V. Manakov, S.P. Novikov, L.P. Pitaevskii,  {\em Theory of solitons, The method of the inverse problem} (Russian), ``Nauka'', Moscow, 1980.
\end{thebibliography}

\title[Soliton asymptotics for the KdV shock problem]{Soliton asymptotics for the KdV shock problem\\ via classical inverse scattering}


\author[I. Egorova]{Iryna Egorova}
\address{B. Verkin Institute for Low Temperature Physics and Engineering\\ 47, Nauky ave\\ 61103 Kharkiv\\ Ukraine}
\email{\href{mailto:iraegorova@gmail.com}{iraegorova@gmail.com}}

\author[J. Michor]{Johanna Michor}
\address{Faculty of Mathematics\\ University of Vienna\\
Oskar-Morgenstern-Platz 1\\ 1090 Wien\\ Austria}
\email{\href{mailto:Johanna.Michor@univie.ac.at}{Johanna.Michor@univie.ac.at}}
\urladdr{\href{http://www.mat.univie.ac.at/~jmichor/}{http://www.mat.univie.ac.at/\string~jmichor/}}

\author[G. Teschl]{Gerald Teschl}
\address{Faculty of Mathematics\\ University of Vienna\\
Oskar-Morgenstern-Platz 1\\ 1090 Wien\\ Austria\\ and Erwin Schr\"odinger International
Institute for Mathematics and Physics\\ Boltzmanngasse 9\\ 1090 Wien\\ Austria}
\email{\href{mailto:Gerald.Teschl@univie.ac.at}{Gerald.Teschl@univie.ac.at}}
\urladdr{\href{http://www.mat.univie.ac.at/~gerald/}{http://www.mat.univie.ac.at/\string~gerald/}}

\keywords{KdV equation,  shock wave, inverse scattering transform,  solitons}
\subjclass[2020]{Primary 37K40, 35Q53; Secondary 37K45, 35Q15}
\thanks{Research supported by the Austrian Science Fund (FWF) under Grant No.\ P31651 and the
National Academy of Sciences of Ukraine (NASU) under Grant No.\  0122U111111.}
\thanks{J. Math. Anal. Appl. {\bf 514}, 126251 (2022)}

\begin{abstract}
We show how the inverse scattering transform can be used as a convenient tool to derive the
long-time asymptotics of the Korteweg--de Vries (KdV) shock problem in the soliton
region. In particular, we improve the results previously obtained via the nonlinear
steepest decent approach both with respect to the decay of the initial conditions as well as
the region where they are valid.
\end{abstract}

\maketitle

\section{Introduction and main results}

Presently the most common method for studying the long-time asymptotic behavior of solutions of completely integrable nonlinear wave equations is 
the nonlinear steepest descent (NSD) method introduced by Deift and Zhou \cite{DZ} extending ideas of Manakov \cite{ma} and Its \cite{its1}.
In particular, this method has superseded the inverse scattering transform (IST) originally used in this context. The purpose of the present note
is to address two issues arising in the context of the NSD method and show how they can be handled, as we feel, more effectively using the IST, at least
in the soliton region. Hence it should be emphasized that the main contribution of this note is not merely the improvement of previous results but the
techniques which lead to these improvements. Clearly these techniques are not limited to our chosen example but apply to other integrable wave equations
solvable via the IST equally well.

The first item is the fact that a common assumption in the context of the NSD method is exponential decay of the initial condition such that the scattering data
allow for an analytic continuation to a neighborhood of the spectrum. Of course this assumption can be weakened by using an analytic approximation of
the scattering data and this was already demonstrated by Deift and Zhou in \cite{DZ}, where Schwartz-type initial conditions were considered. However, it should be pointed out that this technique
is not limited to Schwartz data and can in principle also handle weaker conditions of decay (see \cite{GT} for the case of the KdV equation with decaying initial data). All of this can be considered
well-understood. Nevertheless, working out all steps in full detail requires considerable effort and hence is frequently skipped.

The second item is the fact that the NSD method produces the asymptotics in sectors. While this is of course natural, since the asymptotics are different for different regions,
it still has the drawback that the different sectors usually do not overlap. In particular, we are not aware of an analysis which produces the asymptotics of the KdV
equation, in the simplest case of decaying initial condition, in overlapping domains which cover the entire $x/t$-plane. In this context our second contribution is
that we can establish soliton asymptotics in a regime which is larger than what could previously be covered using the NSD method.

In this vein, the aim of our note is to show the benefits of the classical IST in soliton domains for the study of the KdV equation
\be\label{kdv} 
q_t(x,t) - 6 q(x,t)q_x(x,t)+q_{xxx}(x,t)=0,
\ee
with step-like initial data $q(x)=q(x,0)$ satisfying the condition
\be
\label{ini1}
\lim_{x \to \infty} q(x)= 0, \quad \lim_{x \to -\infty} q(x) = -c^2, \quad c > 0.
\ee
The assumption that the left background $-c^2$ is smaller than the right background $0$ is known as the corresponding shock problem, while the reversed
situation, when the left background is larger than the right background, is known as rarefaction problem.

According to the seminal works by Gurevich  and Pitaevskii (\cite{gp1, gp2}), the long-time asymptotics of solutions of \eqref{kdv}, \eqref{ini1} can
be split into three main regions as illustrated in Figure~\ref{fig:num}:
\begin{itemize}
\item In the left region $x<-6 c^2 t$ the solution is asymptotically close to the background $-c^2$ up to a decaying dispersive tail.
\item In the middle region $-6c^2 t<x<4c^2 t$ the solution is given by a dispersive shock which can asymptotically be described by a slowly modulated elliptic wave.
\item In the right region $4c^2 t < x$ the solution is asymptotically given by a sum of solitons.
\end{itemize}
\begin{figure}
\begin{center}
\includegraphics[width=8cm]{KdVShockSoliton1}
\end{center}
\caption{Numerically computed solution $q(x,t)$ of the KdV equation at time $t=10$, with initial
condition $q(x,0)=\frac{1}{2}(\mathrm{erf}(x)-1)-5\sech(x-1)$.} \label{fig:num}
\end{figure}
Precise formulas for the leading terms of the asymptotic expansion of $q(x,t)$ with respect to $t$ in all three principal regions were derived in \cite{EGKT}  using the vector Riemann--Hilbert approach under the assumption of exponential decay of the initial data to the background constants in \eqref{ini1}. In the soliton region and in the tail region behind the elliptic wave, a complete analysis was given. In the middle region a complete analysis is still open. The asymptotics of $q(x,t)$ in the transition region behind the leading wave front is fully investigated by E.\ Khruslov (\cite{Kh}) by use of the classical IST. He showed that the shock wave splits into a train of asymptotic solitons. The second transition region near the back wave front $\frac x t=-6 c^2$ is not studied yet and the possible behavior of solutions is unknown even on a physical level of rigor.

In the present paper our focus lies on the soliton region $x > 4c^2 t$, where we want to weaken the exponential decay condition
to polynomial decay and, at the same time, cover a larger region. Specifically, we will derive the asymptotics for
\[
x \geq 4c^2t + \frac{m_0-\frac 3 2 -\varepsilon}{2c}\,\log t, \quad \varepsilon >0, \quad m_0 \geq 3,
\]
whereas the asymptotics obtained via NSD in \cite{EGKT} only hold for 
\[x\geq 4c^2 t + \varepsilon t, \quad \varepsilon >0.
\]
Our second aim was to find minimal restrictions on the steplike initial data  for which the 
associated  Riemann--Hilbert problem (RHP) is well-posed. When stating the RHP for a nonlinear equation, 
one assumes that its solution for each $t$ has a behavior that guarantees the existence of Jost solutions. 
In our case, it means that the solution $q(x,t)$ to the initial-value problem \eqref{kdv}, \eqref{ini1} has to have finite first moments of perturbation. If this is not established 
beforehand, then the respective RHP statement is conditional, and one has to prove that the RHP has a solution. 
These minimal restrictions on the initial data to achieve existence of a unique classical solution are the content of Theorem~\ref{thoern}. 

Our main result is the following. 
Denote by $C^n(\R)$ the set of functions $x \in \R \mapsto q(x) \in \R$ which have $n$ continuous derivatives with respect to $x$
and let $C^{n,k}(\R^2)$ be the set of functions $(x,t) \in \R^2 \mapsto q(x,t) \in \R$ which have $n$ continuous derivatives with respect to $x$ 
and $k$ continuous derivatives with respect to $t$. Denote the discrete spectrum of the associated scattering problem by 
$-\kappa_N^2<\dots<-\kappa_1^2$ and the inverse of the norm of the right eigenfunctions by 
$\gamma_j$, $j=1,\dots,N$ (compare Sec.\ \ref{S:scattering}).

\begin{theorem} \label{maintheor1} Let $m_0\geq 3$, $n_0\geq 2$ be arbitrary natural numbers and let $\varepsilon>0$ be arbitrary small. Assume that $q(x)\in C^{n_0}(\R)$ satisfies
\be\label{inithor}\int_0^{+\infty} (1 +|x|^{m_0})\left(\left|\frac{d^n}{d x^n}q(x)\right| +\left|\frac{d^n}{d x^n}(q(-x)+c^2)\right|\right)dx<\infty,\quad  0\leq n\leq n_0.\ee Let $q(x,t)$ be the solution of \eqref{kdv}, \eqref{inithor}. 
Then the asymptotics of $q(x,t)$ in the  region 
\[x \geq 4c^2t + \frac{m_0-\frac 3 2 - \varepsilon}{2c}\,\log t\]  
is the following as $t\to\infty$:
\be \label{asmain}
q(x,t) 
=-\sum_{j=1}^N\frac{2\kappa_j^2}{\cosh^2\big(\kappa_j x - 4\kappa_j^3 t -\frac{1}{2}\log\frac{\gamma_j^2}{2\kappa_j}-\sum_{i=j+1}^N\log\frac{\kappa_j - \kappa_i}{\kappa_i + \kappa_j}\big)} + O\bigg(\frac{1}{t^{m_0-\frac{3}{2}-\varepsilon}}\bigg).
\ee
\end{theorem}

For convenience, we recall the precise result from \cite{EGKT}.

\begin{theorem}[\cite{EGKT}]
Assume that the initial data $q(x)$ of the problem \eqref{kdv}, \eqref{ini1} belongs to $C^7(\R)$ and satisfies
\be\label{ini2}\int_0^{+\infty}\E^{C_0 x}(|q(x)| + |q(-x) + c^2|) dx <\infty, \quad q^{(i)}x^4\in L_1(\R),
\ee
where $i=1,\dots,7$ and $C_0>c>0$. Let $q(x,t)$ be the solution of \eqref{kdv}, \eqref{ini2}. 
Assume that $\delta>0$ is sufficiently small such that the intervals $[4\kappa_j^2 -\delta, 4\kappa_j^2 +\delta]$ are disjoint and lie inside $(4c^2, \infty)$. Then the asymptotics in the soliton region $\frac{x}{t} - 4c^2\geq \varepsilon$ for some small $\varepsilon>0$ are as follows:

If $x\to\infty$, $t\to\infty$ and $\left|\frac{x}{t} - 4\kappa_j^2\right|<\delta$ for some $j$, 
\be\label{as1}
q(x,t)=\frac{-4\kappa_j\gamma_j^2(x,t)}{(1+(2\kappa_j)^{-1}\gamma_j^2(x,t))^2}+O(t^{-l}),\ee
for any $l\in\mathbb N$, where
\be\label{gammaxt}
\gamma_j^2(x,t)=\gamma_j^2\E^{-2\kappa_j x+8\kappa_j^3 t}\prod_{j=j+1}^N\left(\frac{\kappa_i -\kappa_j}{\kappa_i + \kappa_j}\right)^2.
\ee

If  $\left|\frac{x}{t} - 4\kappa_j^2\right|>\delta$ for all $j$, then
$q(x,t)=O(t^{-l})$ for all $l \in\mathbb N$.
\end{theorem}

\begin{remark} Let us check that \eqref{asmain} and \eqref{as1} indeed represent the same leading term 
of the asymptotic expansion. For sufficiently large $t$ the summands in \eqref{asmain} do not interfere up to an exponentially small term. They have nonzero profiles in vicinities of the rays $\frac{x}{t}=4\kappa_j^2$ defined by the phase shifts
\be\label{deltaj}\Delta_j=-\frac{1}{2}\log\frac{\gamma_j^2}{2\kappa_j}-\sum_{i=j+1}^N\log\frac{\kappa_j - \kappa_i}{\kappa_i + \kappa_j}.\ee
On the other hand, using \eqref{gammaxt} and \eqref{deltaj} we get
\begin{align*}
\frac{-4\kappa_j\gamma_j^2(x,t)}{(1+(2\kappa_j)^{-1}\gamma_j^2(x,t))^2}&=-\frac{8\kappa_j^2}{\left(\frac{\sqrt{2\kappa_j}}{\gamma_j(x,t)} +\frac{\gamma_j(x,t)}{\sqrt{2\kappa_j}}\right)^2} = -\frac{8\kappa_j^2}{\left(\E^{\kappa_j x- 4\kappa_j^3 t + \Delta_j} +\E^{-\kappa_j x+4\kappa_j^3 t - \Delta_j}\right)^2},
\end{align*}
 which implies \eqref{asmain}.
\end{remark}
Existence of a unique classical solution of the initial-value problem \eqref{kdv}, \eqref{inithor} considered in Theorem \ref{maintheor1} is provided by the results of Grudsky and Rybkin \cite{GR} which  are applicable for the wider class of initial data essentially bounded from below on the left half axis and decaying with rate $x^{5/2} q(x)\in L_1(\R_+)$ on the right half axis. Recently these results were generalized by 
Laurens \cite{Lau1, Lau2} for initial data, which are the sum of a known steplike KdV solution and an arbitrary function
from $H^{-1}(\R)$. However, all these results do not allow us to control the behavior of $q(x,t)$ as $x\to \pm\infty$ for any fixed $t$, and they do not ensure the requirements on smoothness of the scattering data applied in the  asymptotic analysis. This can be achieved in the framework of the classical IST generalized to the case of steplike solutions.  For steplike KdV solutions the following result holds:

\begin{theorem}[\cite{EGT, ET3}] Let $p_\pm(x)$ be two different algebro-geometric quasi-periodic finite gap potentials and let $p_\pm(x,t)$ be two finite gap KdV solutions associated with the initial data $p_\pm(x,0)=p_\pm(x)$. Assume that $q(x,0)\in C^{n_0}(\R)$ and
\be\label{iniper}\pm \int_0^{\pm\infty}\left|\frac{d^n}{dx^n}(q(x,0) - p_\pm(x))\right|(1 + |x|^{m_0})dx <\infty,\quad 0\leq n\leq n_0,
\ee 
where 
\[ 
m_0\geq 8, \quad \mbox{and}\quad n_0\geq m_0+5,
\]
are some fixed natural numbers. Then there exists a unique classical solution $q(x,t)\in C^{n_0 - m_0 -2,1}(\R^2)$ of the initial-value problem \eqref{kdv}, \eqref{iniper} satisfying
\[ \label{deriv}
\pm\int_0^{\pm\infty}\left|\frac{\partial^n}{\partial x^n}(q(x,t) - p_\pm(x,t))\right| \Big(1 + |x|^{[\frac{m_0}{2}]-2}\Big)dx <\infty,
\quad  0\leq n\leq n_0 - m_0 -2,
\]
and
\[ \label{dert}
\pm\int_0^{\pm\infty}\left|\frac{\partial}{\partial t}(q(x,t) - p_\pm(x,t))\right| \Big(1 + |x|^{[\frac{m_0}{2}]-2}\Big)dx <\infty,
\]
for all $t\in\R$.
\end{theorem}
Hence the largest class of initial data for which unique solvability is established, corresponds to initial datum with $13$ continuous derivatives and $8$ finite moments of perturbation. In our case of constant background solutions
$p_+(x)=p_+(x,t)=0$ and $p_-(x)=p_-(x,t)=-c^2$, we can extend this class 
to initial datum with $7$ continuous derivatives and $4$ finite moments.

\begin{theorem} \label{thoern}  
Given a pair  $(m_0, n_0)\in \N\times\N$ such that
\be\label{estbig}
m_0\geq 4, \quad \mbox{and}\quad n_0\geq m_0+3,\ee assume that  $q(x)\in C^{n_0}(\R)$ and satisfies \eqref{inithor}.
Then there exists a unique classical solution to the initial-value problem \eqref{kdv}, \eqref{inithor}, \eqref{estbig} such that for all $t\in\R$
\be\label{inithor1}\int_0^{+\infty} \Big(1 +|x|^{[\frac{m_0}{2}]-1}\Big)\left(\left|\frac{\partial^n}{\partial x^n}q(x,t)\right| +\left|\frac{\partial^n}{\partial x^n}(q(-x,t)+c^2)\right|\right)dx<\infty,\ee
for all $ 0\leq n\leq n_0-m_0,$ and
\be\label{inithor2}\int_\R \Big(1 +|x|^{[\frac{m_0}{2}]-1}\Big)\left|\frac{\partial}{\partial t}q(x,t)\right| dx<\infty.
\ee
\end{theorem}
Thus condition \eqref{estbig} ensures the well-posedness of the  RHP which forms the base of the 
long-time asymptotic analysis of steplike solutions via the NSD approach. It also secures the smoothness and decay 
properties of the scattering data required for the asymptotic analysis used in Theorem~\ref{maintheor1}.

The results of Theorem~\ref{thoern} can be applied to study the RHP of low regularity associated with the KdV shock problem.
These problems of low regularity for the KdV and mKdV equations were studied until now only for fast decaying 
initial data, see \cite{DZ, GT, Len18, Lenells}.  

The proof of Theorem \ref{thoern} is based on the well-developed (direct/inverse) scattering theory for the Schr\"{o}dinger operator with steplike potential (cf.\ \cite{BF, CK, EGLT}), where a one-to-one correspondence is established between certain classes of potentials and classes of the scattering data. The left and right Marchenko equations, which connect the transformation operators and the generalized Fourier transforms of the scattering data, are the key ingredient in the solution of the inverse scattering problem. Moreover, the decay rates and the smoothness of the kernels of the Marchenko equations directly determine the decay rates
of their solutions, and therefore the rate of convergence of the potential to its asymptotes. We list characteristic properties of the scattering data for the potential of the class \eqref{inithor}, \eqref{estbig} in Section \ref{S:scattering}. Assuming that the solution of the Cauchy problem \eqref{kdv}, \eqref{inithor}, \eqref{estbig} exists in a class of functions which tend to their background asymptotes at least slowly, we show the evolution of the scattering data with respect to time and obtain the time-dependent Marchenko equations. For steplike initial data it was first done in \cite{Kh}, and we recall these results in the beginning of Section~\ref{Cauchypr}.  However, the corresponding characteristic properties might not be preserved by the time evolution. So the main body of Section~\ref{Cauchypr} consists of two parts. In Subsection~\ref{subsection1} we compute the decay rate of the right Marchenko equation kernel provided that the initial datum satisfies \eqref{inithor}, \eqref{estbig}, and in Subsection~\ref{subsection2} we do the same for the left kernel. The main losses in the rate of decay $m_0 \mapsto [\frac{m_0}{2}]$ occur precisely here. The constant backgrounds allow us to get more precise estimates compared to those given in \cite{EGT, ET3} for finite gap backgrounds. Finally, in Section \ref{sec4} we analyze the contribution of each of the three summands of the right kernel of the Marchenko equation on the asymptotics in the soliton region and prove Theorem~\ref{maintheor1}.


\section{Scattering theory for the Schr\"{o}dinger operator}  \label{S:scattering}

In this section we briefly describe the characteristic (necessary and sufficient) properties of the scattering data for the Schr\"{o}dinger operator
\[ (Lf)(x):=-\frac{d^2}{dx^2}f(x) + q(x)f(x)\] with steplike potential \eqref{ini1} satisfying \eqref{inithor}, assuming that $m_0\geq 1$ and $n_0\geq 0$. Let $\la$ be the spectral parameter of the problem 
\be\label{sch} -\frac{d^2}{dx^2}f(x) + q(x)f(x)=\la f(x).\ee
We introduce additional spectral parameters $k=\sqrt\la$ and $k_1=\sqrt{\la + c^2}$ which map the domains $\C\setminus \R_+$ and $\mathcal D:=\C\setminus [-c^2, \infty)$ conformally onto $\C^+$. The boundary of the domain $\mathcal D$ is treated as consisting of two sides of the cut along the interval $[-c^2, \infty)$, with distinguished points $\la^u=\la+\I 0$ and $\la^l=\la- \I 0$ on this boundary. We assume that $\la\in\overline {\mathcal D}$. This gives a one-to-one correspondence between the parameters $k$, $k_1$ and $\la$. 

The spectrum of the operator $L$ consists of a continuous spectrum of multiplicity two on $\R_+$, a continuous spectrum of multiplicity one on $[-c^2, 0]$ and a finite discrete spectrum $\{\la_1, \dots, \la_N\}$, where $\la_N<\dots<\la_1<-c^2.$

According to \cite{M} and \cite{DT}, equation \eqref{sch} has two Jost solutions
\[\aligned \phi(\la,x)&=\E^{\I k x}\left(1 +\int_0^\infty B(x,y)\E^{2\I k y} dy\right), \quad k\in\ol{\C^+},  \\ \phi_1(\la,x)&=\E^{-\I k_1 x}\left(1 +\int^0_{-\infty} B_1(x,y)\E^{-2\I k_1 y} dy\right),\quad k_1\in\ol{\C^+}, 
\endaligned
\]
where $B(x,y)$ and $B_1(x,y)$ are real-valued functions (the kernels of the transformation operators) satisfying
the properties
\be\label{form1}
B(x,0)=\int_x^{+\infty} q(y) dy,\quad B_1(x,0)=\int_{-\infty}^x (q(y)+c^2)dy.\ee
One of the main characteristics of the spectral problem \eqref{sch}, \eqref{inithor} is the Wronskian
$W(\la):=W(\phi(\la,\cdot), \phi_1(\la,\cdot))$ of the Jost solutions, where 
$W(f,g)=f(x)g^\prime(x) - g(x)f^\prime(x)$. It is well-known that $W(\la)$ has simple zeros at 
the points of the discrete spectrum $\la_j$ and possibly at the end of the continuous spectrum, but no other zeros. 
If $W(-c^2)=0$ we call the point $-c^2$ a resonant point. Denote by
\[\gamma_j^2:=\frac{1}{\int_{\R}\phi^2(\la_j,x) dx},\quad \gamma_{j,1}^2:=\frac{1}{\int_{\R}\phi_1^2(\la_j,x) dx}\]
the right, left normalizing constants, respectively.


Consider the standard  scattering relations
\[\aligned T(\la)\phi_1(\la, x)&=\ol{\phi(\la,x)} + R(\la)\phi(\la,x),\quad k\in\R,\\
T_1(\la)\phi(\la, x)&=\ol{\phi_1(\la,x)} + R_1(\la)\phi_1(\la,x),\quad k_1\in\R,\endaligned\]
where the transmission and reflection coefficients constituting the entries of the scattering matrix are 
defined as usual
\be\aligned T(\la)&:=\frac{W(\ol{\phi(\la)}, \phi(\la))}{W(\la)},\quad R(\la):=\frac{W(\ol{\phi(\la)}, \phi_1(\la))}{W(\la)}, \quad \quad k\in\R,\\
T_1(\la)&:=-\frac{W(\ol{\phi_1(\la)}, \phi_1(\la))}{W(\la)},\quad R_1(\la):=-\frac{W(\ol{\phi_1(\la)}, \phi(\la))}{W(\la)}, \quad k_1\in\R.
\endaligned\ee

\begin{lemma}[\cite{BET, CK, EGLT}] \label{1-3}
Let $q(x)$ satisfy \eqref{inithor} with $m_0\geq 1$ and $n_0\geq 0$.
The entries of the scattering matrix satisfy
\vskip 3mm
\begin{itemize}
\item[{\bf I.(a)}] $T(\lambda+\I 0) =\overline{T(\lambda-\I 0)}$ and
$R(\lambda+\I 0) =\overline{R(\lambda-\I 0)}$ for $k(\la)\in\R$; \newline
$T_1(\lambda+\I 0) =\overline{T_1(\lambda-\I 0)}$ and 
$R_1(\lambda+\I 0) =\overline{R_1(\lambda-\I 0)}$ for $k_1(\la)\in\R$.
\vskip 1mm
\item[{\bf (b)}] $\dfrac{T_1(\lambda)}{\overline{T_1(\lambda)}}= R_1(\lambda)$ for $k_1(\la)\in [-c,c]$.
\item[{\bf (c)}] $1 - |R(\lambda)|^2 =
\dfrac{k_1}{k}\,|T(\lambda)|^2$ and $1 - |R_1(\lambda)|^2 =
\dfrac{k}{k_1}\,|T_1(\lambda)|^2$ for $k(\lambda)\in\R$.
\item[{\bf (d)}] $\overline{R(\lambda)}T(\lambda) +
R_1(\lambda)\overline{T(\lambda)}=\overline{R_1(\lambda)}T_1(\lambda) +
R(\lambda)\overline{T_1(\lambda)}=0$ for $k(\lambda)\in \R$.
\vskip 1mm
\item[{\bf (e)}] $T(\lambda) = 1 +O(\frac{1}{\sqrt\la})$, $T_1(\lambda) = 1 +O(\frac{1}{\sqrt\la})$  as $\lambda\to\infty$, \\
 $R(\lambda) = O(\frac{1}{\sqrt\la})$, $R_1(\lambda) = O(\frac{1}{\sqrt\la})$ as $\lambda\to\infty$.
\vskip 1mm
\item[{\bf II. \ }]
The functions $T(\lambda)$ and $T_1(\la)$ can be analytically continued to 
$\mathcal D$ satisfying
\[ 
2\I k(\la) T^{-1}(\la)  = 2\I k_1(\la) T_1^{-1}(\la) =:W(\la),
\]
where $W(\la)$ has the following properties: \begin{itemize}
\item[$(i)$] It is holomorphic in the domain $\mathcal D$ and continuous up to the boundary. Moreover, $W(\la + \I 0)=\ol{W(\la - \I 0)}\neq 0$ as $\la \in (-c^2, \infty)$.
\item[$(ii)$] In $\mathcal D$ its only zeros are $\la_1, \dots, \la_N$, and
\[ 
\left(\frac{d W}{d \la}\,(\la_j)\right)^{-2}=\gamma_j^2\gamma_{j,1}^2.
\]
\item[$(iii)$] If $W(-c^2)=0$ then 
\[ 
W(\la)=\I\gamma\sqrt{\la +c^2}(1+o(1)), \quad \mbox{as}\ \ \la\to -c^2,\quad \mbox{where}\ \ \gamma\in\R\setminus\{0\}.
\]
\end{itemize}
\item[{\bf III. }] $R(\la)$  and $R_1(\la)$ are continuous for $k(\la)\in\R$ and $k_1(\la)\in\R$, respectively. 
\end{itemize}
\end{lemma}
Define
\[K(x,y)=\frac{1}{2} B\Big(x, \frac{y-x}{2}\Big), \quad K_1(x,y)=\frac{1}{2} B_1\Big(x, \frac{y-x}{2}\Big),\]
and
\be\label{defF}
\aligned F(x)&=\frac{1}{2\pi}\int_{\R} R(\la)\E^{\I k x} dk +\frac{1}{4\pi}\int_{-c^2}^0 |T_1(\la)|^2\E^{\I k x}\frac{d\la}{|k_1|} +\sum_{j=1}^N\gamma_j^2\E^{-\kappa_j x};\\
F_1(x)&=\frac{1}{2\pi}\int_{\R} R_1(\la)\E^{-\I k_1 x}d k_1 +\sum_{j=1}^N\gamma_{j,1}^2 \E^{\kappa_{j,1} x},\quad \kappa_{j,1}=\sqrt{-c^2-\la_j}>0.\endaligned\ee
These functions are connected by the right and left Marchenko equations (\cite{BF, CK})
\be\label{March}
\aligned K(x,y) + F(x+y) + \int_x^{+\infty} K(x,s) F(y+s)ds&=0,\quad y>x;\\
K_1(x,y) + F_1(x+y) + \int^x_{-\infty} K_1(x,s) F_1(y+s)ds&=0,\quad y<x.
\endaligned\ee

\begin{lemma}[\cite{M}] \label{lemma 2.8} Let $q(x)$ satisfy \eqref{inithor} with $m_0\geq 1$ and $n_0\geq 0$. Then 
\vskip 2mm
{\bf IV.}\ $F(x), F_1(x)\in C^{n_0+1}(\R)$ and
\[ \label{dec} 
\int_0^{+\infty} \left|\frac{d^n F}{d x^n}\right|(1 +|x|^{m_0}) dx<\infty, \quad 
\int^0_{-\infty} \left|\frac{d^n F_1}{d x^n}\right|(1 +|x|^{m_0}) dx<\infty, \quad 1\leq n\leq n_0+1.
\]
\end{lemma}
The set of scattering data for the operator $L$ can be defined as follows:
\be\label{scatdata}\aligned
\mathcal S(m_0, n_0):=&\big\{R(\la), T(\la), k\in\R;\  R_1(\la), T_1(\la), k_1\in \R; \\ & \quad \la_1, \dots,\la_N\in (-\infty,-c^2),\ \gamma_1^2, \gamma_{1,1}^2, \dots, \gamma_N^2, \gamma_{N,1}^2\big\}.
\endaligned\ee
The functions $F(x)$ and $F_1(x)$ are uniquely defined by this set. 
\begin{theorem}[\cite{EGLT}] \label{charak}
Properties {\bf I}--{\bf IV} of Lemmas~\ref{1-3} and \ref{lemma 2.8} are necessary and sufficient for 
$\mathcal S(m_0, n_0)$ to be the set of scattering data for the Schr\"odinger operator $L$ with potential $q(x)$ satisfying \eqref{inithor} for any $m_0\geq 1$ and $n_0\geq 0$.
\end{theorem}
The sufficiency of these properties is established as follows. Given arbitrary $m_0\geq 1$ and $n_0\geq 0$ and the set $\mathcal S(m_0, n_0)$ consisting of four functions and 3N numbers as in \eqref{scatdata}, assume that this set satisfies properties {\bf I} to {\bf IV}. Then the Marchenko equations \eqref{March} are uniquely solvable (\cite{M}) for the kernels of the transformation operators $K$ and $K_1$. Moreover, the functions (cf.\ \eqref{form1})
\[ q_+(x)=-2\frac{d}{dx} K(x,x)=-\frac{d}{dx} B(x,0), \quad q_-(x)=2\frac{d}{dx} K_1(x,x)=\frac {d}{dx} B_1(x,0),\]
are such that $q_\pm\in C^{n_0}(\R)$ and (\cite{M})
\[\int_{\R_\pm}\left|\frac{d^n}{dx^n} q_\pm(x)\right|(1+|x|^{m_0})dx<\infty, \quad 0\leq n\leq n_0.\]
The last and most important step establishes that $q_+(x)=q_-(x) - c^2:=q(x)$ (cf.\ \cite{EGLT}), that is, $q(x)$ satisfies \eqref{inithor}. Moreover, the  operator $L$ with potential $q(x)$ reconstructed by use of the Marchenko equations, 
has the chosen set $\mathcal S(m_0, n_0)$ as the set of scattering data.
This scattering theory forms the basis of the inverse scattering transform which we apply in the next section to prove Theorem \ref{thoern}.

\section{The Cauchy problem for the KdV equation with steplike initial data of the class (\ref{inithor}), (\ref{estbig})}\label{Cauchypr}
We first recall some well-known facts about the Lax pair. By a classical solution of the KdV equation we mean a solution that has $3$ continuous derivatives with respect to $x$ and one with respect to $t$. Let $v(x,t)$ be such a solution. Introduce the Lax operators
\[L_v(t)=-\pa_x^2 + v(x,t), \quad P_v(t)=-4\pa_x^3 + 6 v(x,t)\pa_x + 3v_x(x,t).\] 
As is known (\cite{Lax}), the KdV equation is equivalent to the Lax equation
\[\pa_t L_v(t)=[P_v(t),\ L_v(t)]\]
considered in $H^5(\R)$. It implies the unique solvability  of the compatibility system
\be\label{comp} L_v(t)u=\la u, \quad u_t= P_v(t)u,\ee for any initial condition $u(\la,0,0)$ and $u_x(\la, 0,0)$. In turn,  if $u_1$ and $u_2$ are two solutions of \eqref{comp}, then their Wronskian does not depend either on $x$ or on $t$. Let $\psi_\pm(\la, x,t)$  be two Weyl solutions of the equation $L_v(t)u=\la u$ normalized by the condition $\psi_\pm(\la,0,t)=1$ and let $m_\pm(\la,t)=\frac{\pa}{\pa x}\psi_\pm(\la,0,t)$ be the Weyl functions.
\begin{lemma}[\cite{EGT}] Set
\[
\alpha_\pm(\la,t)=\exp\left(\int_0^t\Big(2\big(v(0,s) + 2\la\big)m_\pm(\la,s) - v_x(0,s)\Big)ds\right).
\]
Then the functions $u_\pm(\la,x,t)=\alpha_\pm(\la,t)\psi_\pm(\la,x,t)$ solve  \eqref{comp}.
\end{lemma}
In particular, the function $u(\la,x,t)=\E^{\I k x + 4 \I k^3 t}$ solves \eqref{comp} with $v(x,t)=0$ and the function $u_1(\la,x,t)=\E^{-\I k_1 x - 4 \I k^3_1 t + 6\I c^2 k_1 t}$ solves \eqref{comp} with $v(x,t)=-c^2$.
For these solutions, 
\be\label{al}
\alpha_+(\la,t)=:\alpha(\la,t)=\E^{4\I k^3 t}, \quad \alpha_-(\la,t)=:\alpha_1(\la,t)=\E^{- 4 \I k^3_1 t + 6\I c^2k_1 t}.
\ee
These properties allow us to derive the evolution  of the scattering data with respect to time. Indeed,
assume that there exists a classical solution $q(x,t)$ of the Cauchy problem \eqref{kdv}, \eqref{inithor} for some 
pair $(m_0, n_0)$ and assume that $q(x,t)$ has at least a finite first moment of perturbation, that is,
\be\label{thor1}\int_0^{+\infty} (1 +|x|)\left(\left|\frac{\pa^n}{\pa x^n}q(x,t)\right| +\left|\frac{\pa^n}{\pa x^n}(q(-x,t)+c^2)\right|\right)dx<\infty,\quad  0\leq n\leq 3.\ee
Then we can apply the results of the previous section to the time-dependent Schr\"{o}\-dinger operator $L_q(t)$. In particular, one can construct the Jost solutions of the equation $-\pa_x^2 y + q(x,t)y=\la y$ by 
\be\label{jostdep}\aligned
\phi(\la,x,t)&=\E^{\I k x} + \int_x^{+\infty}K(x,y,t)\E^{\I k y} dy,\\
\phi_1(\la,x,t)&=\E^{-\I k_1 x} + \int^x_{-\infty}K_1(x,y,t)\E^{-\I k_1 y} dy,\endaligned\ee
and introduce their time-dependent Wronskian $W(\la,t)$ and the scattering relations,
\[\aligned T(\la,t)\phi_1(\la, x,t)&=\ol{\phi(\la,x,t)} + R(\la,t)\phi(\la,x,t),\quad k\in\R,\\
T_1(\la,t)\phi(\la, x,t)&=\ol{\phi_1(\la,x,t)} + R_1(\la,t)\phi_1(\la,x,t),\quad k_1\in\R.
\endaligned\]
The left/right normalizing constants now also depend on $t$ and are given by
\[\gamma_j^2(t):=\frac{1}{\int_{\R}\phi^2(\la_j,x,t) dx},\quad \gamma_{j,1}^2(t):=\frac{1}{\int_{\R}\phi_1^2(\la_j,x,t) dx}.\]
Since $\alpha(\la,0)=\alpha_1(\la,0)=1$  by \eqref{al}, it follows from the considerations above that 
\be\label{dynn}\aligned W(\la,t)&=\frac{W(\la,0)}{\alpha(\la,t)\alpha_1(\la,t)},\quad T_1(\la,t)=T_1(\la,0)\alpha(\la,t)\alpha_1(\la,t), \\
R_1(\la,t)&=R_1(\la,0)\alpha_1^2(\la,t),\quad R(\la,t)=R(\la,0)\alpha^2(\la,t),\endaligned
\ee 
and the time evolution of the scattering data is therefore given by (cf.\ \cite{Kh, EGT})
\begin{equation}
\begin{split}
\label{evol}
R(\la,t)&=R(\la)\E^{8\I k^3 t}, \\
T(\la,t)&= T(\la)\E^{4\I k^3 t- 4 \I k^3_1 t + 6\I c^2k_1 t}, \\
\gamma_j^2(t)&=\gamma_j^2 \E^{8 \kappa_j^3 t},
\end{split}
\quad 
\begin{split}
R_1(\la,t)& = R_1(\la)\E^{- 8 \I k^3_1 t + 12\I c^2k_1 t}, \\ 
T_1(\la,t)&=T_1(\la)\E^{4\I k^3 t- 4 \I k^3_1 t + 6\I c^2k_1 t}, \\ 
\gamma_{j,1}(t)&=\gamma_{j,1}\E^{-(8\kappa_{j,1}^3 + 12 c^2\kappa_{j,1})t}.
\end{split}
\end{equation}
Here we have abbreviated $R(\lambda)=R(\lambda,0)$, etc.
On the spectrum of multiplicity one, 
\[ |T_1(\la,t)|^2 = |T_1(\la)|^2\E^{8\I k^3 t}, \quad \la\in [-c^2,0]\pm\I 0.
\]
Therefore the time-dependent Marchenko equations have the form
\be\label{March1}
\aligned K(x,y,t) + F(x+y,t) + \int_x^{+\infty} K(x,s,t) F(y+s,t)ds&=0,\quad y>x,\\
K_1(x,y,t) + F_1(x+y,t) + \int^x_{-\infty} K_1(x,s,t) F_1(y+s,t)ds&=0,\quad y<x,\endaligned
\ee
where
\be\label{defF1}
\aligned F(x,t)=&\frac{1}{2\pi}\int_{\R} R(\la)\E^{\I k x+8\I k^3 t} dk +\frac{1}{2\pi}\int_{-c^2}^0 \frac{k|T_1(\la)|^2}{|k_1|}\E^{\I k x+8\I k^3 t}d k +\sum_{j=1}^N\gamma_j^2\E^{-\kappa_j x+8\kappa_j^3 t},\\
F_1(x,t)=&\frac{1}{2\pi}\int_{\R} R_1(\la)\E^{-\I k_1 x-8\I k_1^3 t + 12\I c^2 k_1 t}d k_1 
+\sum_{j=1}^N\gamma_{j,1}^2 \E^{\kappa_{j,1} x-8\kappa_{j,1}^3t -12 c^2\kappa_{j,1}t}.
\endaligned
\ee
Formulas \eqref{evol}--\eqref{defF1} were obtained assuming that $q(x,t)$ exists and tends to the background 
constants as in \eqref{thor1}. To prove that $q(x,t)$ indeed exists and is unique we have to check that the set
\be\label{scatdatat}\aligned
\mathcal S(t):=&\big\{R(\la,t), T(\la,t), k\in\R;\  R_1(\la,t), T_1(\la,t), k_1\in \R; \\ & \quad
\la_1, \dots, \la_N\in (-\infty,-c^2), \gamma_1^2(t), \gamma_{1,1}^2(t), \dots, \gamma_N^2(t), \gamma_{N,1}^2(t)\big\}
\endaligned\ee
defined via \eqref{evol}--\eqref{defF1} and \eqref{scatdata} (corresponding to the initial data) satisfies the 
necessary and sufficient conditions {\bf I}--{\bf IV}. It may happen that  $\mathcal S(t)$ will have less moments and derivatives 
than the initial data, that is, $\mathcal S(t)=\mathcal S(m_0(t), n_0(t))$ with $1\geq m_0(t)\geq m_0$, $3\geq n_0(t)\geq n_0$. This however will still guarantee unique solvability of \eqref{March1} and the equality $q(x,t)\equiv q_1(x,t)$, 
where
\[ q(x,t)=-2\frac{d}{dx}K(x,x,t),\quad q_1(x,t)=2\frac{d}{dx} K_1(x,x,t) - c^2, \quad x\in\R,
\]
as well as the required decay of these functions to their backgrounds.
Let us first check properties {\bf I}--{\bf III}.  Properties {\bf I.(a), (b)} follow immediately from 
\be\label{zum}
\aligned\alpha(\la+\I 0, t)& =\ol{\alpha(\la-\I 0, t)}=\alpha^{-1}(\la+\I 0, t), \quad  \la\in [0,\infty);\\ \alpha_1(\la+\I 0, t) &=\ol{\alpha_1(\la-\I 0, t)}=\alpha^{-1}_1(\la+\I 0, t), \quad \la\in[-c^2, \infty);\\ 
 \alpha(\la-\I 0, t)& =\alpha(\la+\I 0, t)\in\R, \quad \la\in [-c^2,0].\endaligned\ee
Since $|\alpha(\la,t)|=|\alpha_1(\la,t)|=1$ as $\la\in\R_+$, then the moduli of the entries of the scattering matrix do not depend on time on the spectrum of multiplicity 2, i.e. when $k\in\R$. This proves {\bf I.(c)}. Property {\bf I.(d)} follows from \eqref{dynn} and \eqref{zum}. Finally, for $k\to\infty$
\begin{align*}
\log(\alpha(\la,t)\alpha_1(\la,t))&=4\I (k^3 - k_1^3)t +6\I c^2k_1 t=O(k^{-1}),\\ 
\alpha(\la,t)\alpha_1(\la,t)&= 1+O(k^{-1}),
\end{align*}
which proves {\bf I.(e)}.
To check {\bf II},  note that $\alpha(\la,t)$ and $\alpha_1(\la,t)$ are holomorphic in $\C\setminus [-c^2, \infty)$ and continuous up to the boundary. This proves $(i)$. For $(ii)$ we use 
\[\frac{\pa W(\la,t)}{\pa \la}\Big|_{\la=\la_j}=(\alpha(\la_j,t)\alpha_1(\la_j,t))^{-1}\frac{d W(\la)}{d\la}\Big|_{\la=\la_j},
\] 
which is true since $W(\la_j)=0$. Together with \eqref{evol} this impies $(ii)$,
\[
\left(\frac{\pa W}{\pa \la}\,(\la_j,t)\right)^{-2}=\gamma_j^2(t)\gamma_{j,1}^2(t).
\]
Property {\bf II}, $(iii)$ follows from $\alpha(-c^2,t)\alpha_1(-c^2,t)\in\R\setminus\{0\}$.

The main difficulty in the application of IST to steplike solutions arises in the verification 
of property {\bf IV} and in finding optimal constants $m_0(t), n_0(t)$, which actually do not depend on $t$ as we will see. 

First of all, to differentiate $F(x,t)$ and $F_1(x,t)$ with respect to $x$ we need more decay of the reflection coefficients than established in 
{\bf I.(e)} of Lemma \ref{1-3}. But $R(\la)$ and $R_1(\la)$ are the reflection coefficients of the potential satisfying \eqref{inithor}, \eqref{estbig} and 
for such a potential it was proven in \cite{EGLT} that 
as $\la \to \infty$,
\[
 \frac{d}{dk}R(\la)=g_{+,s}(\la)\la^{-\frac{n_0+1}{2}}, \quad  \frac{d}{dk_1}R_1(\la)=g_{-,s}(\la)\la^{-\frac{n_0+1}{2}},\quad s=0, \dots,m_0-1,
 \]
 where $\sqrt\la\cdot g_\pm(\la)\in L^2 (a, +\infty)$ and $a$ is a sufficiently large positive number.

Let us examine the behavior of $F(x, t)$ as $x \rightarrow +\infty$ and $F_1(x, t)$ as $x \rightarrow -\infty$.
The summands which correspond to the discrete spectrum are exponentially small with respect to large 
$x$ for a fixed $t$, so they do not contribute to the behavior of the kernels. 

\subsection{Asymptotic behavior of the right kernel $F(x, t)$.} \label{subsection1}
Denote the two summands of $F(x, t)$ in \eqref{defF1} containing the integrals by $F_R(x, t)$ and $F_T(x, t)$. 
Then $F_T(x, t)$ can be rewritten as
\[
F_T(x, t) = \dfrac{1}{4\pi} \int_{-c^2}^0 |T_1(\la)|^2 k_1^{-1} e^{8ik^3t}e^{ikx}d\la = \dfrac{1}{\pi}\int_c^0 P(h) \psi(h) e^{-hx} dh,
\]
where $\sqrt{\la} = k = \I h$, $\psi(h)=e^{8h^3t}$ and
\begin{equation} \label{eq3.19}
P(h) = \dfrac{ \I h W(\phi_1(\la, \cdot, 0), \ol{\phi_1(\la, \cdot, 0)})}{W(\phi_1(\la, \cdot, 0), \phi(\la, \cdot, 0)) W(\ol{\phi_1(\la, \cdot, 0)}, \phi(\la, \cdot, 0))}.
\end{equation}
Using the symmetry property $R(\la(k))=\ol{R(\la(-k))}$ we can write
\begin{equation*}
F_R(x, t) = \dfrac{1}{\pi} \re \int_0^{+\infty}R(k)\psi(- \I k) e^{\I kx}dk,
\end{equation*}
where $R(k): = R(\la(k), 0)$ and $k \geq 0$.
Note that $P(h)$ does not depend on $t$, and $P(h)<0$ when $h \in (0, c)$. Since time $t$ is fixed, we omit $t$ in the notation of $\psi(h)$.
Let us abbreviate $\psi_s^{(j)}=\frac{\partial^j}{\partial s^j} \psi(h(s), t)$ and $\psi_s' = \frac{\partial}{\partial s} \psi(h(s), t)$.
Evidently,
\begin{equation} \label{eq3.20}
\psi_h^{(j)}(0) = \I^j \psi_k^{(j)}(0) \in \R.
\end{equation}
Integrating $F_R(x, t)$ by parts yields
\begin{equation*}
\begin{aligned}
\re \int_0^{\infty} R(k)\psi(-\I k)e^{\I kx}dk = \re \bigg\{-\dfrac{1}{\I x}R(0)\psi(0) + \dfrac{1}{(\I x)^2}(R_k' \psi + R \psi_k^\prime)(0) \\
+ \dots +  \frac{(-1)^m}{(ix)^m}\dfrac{\partial^{m-1}(R\psi)}{\partial k^{m-1}}(0) + \frac{(-1)^m}{(\I x)^m} \int_0^{\infty}\dfrac{\partial^{m}(R\psi)}{\partial k^{m}}e^{\I kx}dk \bigg\}.
\end{aligned}
\end{equation*}
We split $F_T(x, t)$ into a sum of two integrals, $\int_0^{\frac{c}{2}}  + \int_{\frac{c}{2}}^c$.
The second integral is evaluated as $O(e^{-\frac{c}{2}x})$, and this estimate can be differentiated with respect to 
$x$, which is sufficient for our purpose. On the other hand, $\phi_1(\la,x,t)$ does not have bounded derivatives with respect to $h$ at $h=c$, therefore $|T_1(\la,t)|^2$ is not differentiable too. Integrating  the first integral by parts yields for 
$\pi F_T(x, t)$
\begin{equation*}
\begin{aligned}
\int_c^0 P(h) \psi(h) \E^{-hx} dh &= - \frac{1}{x}P(0)\psi(0) - \frac{1}{x^2}(P_h'\psi + P\psi_h')(0) - \dots  
\\
& \quad - \frac{1}{x^m}\dfrac{\partial^{m-1}(P\psi)}{\partial h^{m-1}}(0) + \dfrac{1}{x^m} \int_{\frac{c}{2}}^0 \dfrac{\partial^{m}(P\psi)}{\partial h^{m}}e^{-hx}dh + O(\E^{-\frac{c}{2}x}),
\end{aligned}
\end{equation*}
where the term $O(e^{-\frac{c}{2}x})$ contains the second integral and integrands of the first integral 
corresponding to $h = \frac{c}{2}$.

Our next aim is to prove that
\begin{equation} \label{eq3.21}
\lim_{k\rightarrow +0} \bigg\{ \I^{j+1} \dfrac{\partial^{j}(R\psi)}{\partial k^{j}}(k) \bigg\} = \lim_{h\rightarrow+0} \dfrac{\partial^{j}(P\psi)}{\partial k^{j}}(h).
\end{equation}
The limit on the right side is taken from the side of the spectrum of multiplicity one. 
Since $\psi \in C^{\infty}(\R)$, using \eqref{eq3.20} we get
\begin{equation*}
\dfrac{\partial^{j}(P\psi)}{\partial k^{j}}(+0) = \sum_{s=0}^j C_j^s P_h^{(j-s)}(+0)\I^s \psi_k^{(s)}(0).
\end{equation*}
Here $C_j^s$ are the binomial coefficients. Compare this formula with 
\begin{equation*}
\re \bigg\{ \I^{j+1} \dfrac{\partial^{j}(R\psi)}{\partial k^{j}}(+0) \bigg\} = 
\sum_{s=0}^j C_j^s \I^{j+1-s} R_k^{(j-s)}(+0)\I^s \psi_k^{(s)}(0),
\end{equation*}
and taking into account that $\I^s \psi_k^{(s)}(0) \in \R$, we see that  (\ref{eq3.21}) is equivalent to
\begin{equation} \label{eq3.22}
\lim_{k\rightarrow +0} \re\bigg\{ \I^{l+1}R_k^{(l)}(k)\bigg\} = \lim_{h\rightarrow+0} P_h^{(l)}(h).
\end{equation}
To prove this, recall that $\phi_1(x, \la, 0)$ and $\overline{\phi_1(x, \la, 0)}$ are $m_0$-times differentiable by $k$ in a neighborhood of the point $k=0$. The function
 $\phi(x, \la, 0)$ has $m_0-1$ derivatives in $k$ at $k = 0$. We define $P_1(k) = P(-\I k)$, where $P$ is determined by (\ref{eq3.19}). The function $P_1(k)$ is defined on the right side of $[0, ic]$, but can be redefined by the same formula to the right of the point $k=0$. Moreover, it has $m_0-1$ continuous derivatives at $k=0$, namely
\begin{equation*}
\lim_{k\rightarrow+0} \I^l \dfrac{d^l P_1(k)}{dk^l} = \lim_{h\rightarrow+0} \dfrac{d^lP}{dh^l}.
\end{equation*}
Now we extend $P_1$ for $k\geq 0$. We can see that
\begin{equation*}
P_1(k) = \dfrac{\I W(\phi_1, \ol{\phi_1})(-2\I k)}{2W(\phi_1, \phi) W(\ol{\phi_1}, \phi)} = \dfrac{\I W(\phi_1, \ol{\phi_1})W(\phi, \overline{\phi})}{2W(\phi_1, \phi) W(\ol{\phi_1}, \phi)}.
\end{equation*}
Substituting
\begin{equation*}
\phi = \dfrac{W(\phi, \overline{\phi_1)}}{W(\phi_1, \overline{\phi_1})}\phi_1 + \dfrac{W(\phi, \phi_1)}{W(\phi_1, \overline{\phi_1})} \ol{\phi_1}
\end{equation*}
in the numerator of the previous fraction we get
\begin{equation} \label{eq3.23}
P_1(k) = \dfrac{\I}{2} \left( -\dfrac{W(\phi_1, \ol{\phi})}{W(\phi_1, \phi)} + \dfrac{W(\ol{\phi_1}, \ol{\phi})} {W(\ol{\phi_1}, \phi)} \right)=
\dfrac{\I}{2} \left( -\dfrac{V(k)}{W(k)} + \frac{\ol{W(k)}}{\ol{V(k)}}\right),
\end{equation}
where
\begin{equation*}
W(k) = W(\phi_1, \phi), \qquad
V(k) = W(\phi_1, \ol{\phi}).
\end{equation*}
This representation is only possible for $k \geq 0$ because $\ol{\phi(\la, x, 0)}$ cannot be continued to $[0, \I c]$.

\begin{lemma}
The following equality is valid,
\begin{equation} \label{eq3.24}
V_k^{(s)} (0) = (-1)^s W_k^{(s)} (0), \qquad  s= 0, \dots, m_0 -1.
\end{equation}
\end{lemma}
\begin{proof}
Denote by  $f_n(x)$  any function such that $f_n(x) \in C^{m_0-1}[0, \epsilon)$.
Since
\begin{equation*}
\phi(\la, x) + \ol{\phi(\la, x)} = 2 \cos(kx) + 2\int_x^{\infty} K(x, y)\,\cos (ky) dy,
\end{equation*}
then
\begin{equation*}
\phi(\la, x) + \ol{\phi(\la, x)} = f_1(k^2), \quad \dfrac{\partial}{\partial x}(\phi(\la, x) + \ol{\phi(\la, x)}) = f_2(k^2).
\end{equation*}
Also $\phi_1(\la, 0)=f_3(k^2)$ and $\tfrac{\partial}{\partial x}\phi_1(\la, 0)=f_4(k^2)$.
On the other hand,
\begin{equation*}
\phi(\la, x) - \ol{\phi(\la, x)} = 2\I k \left( \dfrac{\sin kx}{k} + \int_x^{\infty} K(x, y) \dfrac{\sin ky}{k} dy\right),
\end{equation*}
thus
\begin{equation*}
\phi(\la, x) - \ol{\phi(\la, x)} = kf_5(k^2), \quad \dfrac{\partial}{\partial x}(\phi(\la, x) - \ol{\phi(\la, x)}) = kf_6(k^2).
\end{equation*}
Therefore
\begin{equation*}
V(k) + W(k) = f_7(k^2), \qquad V(k) - W(k) = f_8(k^2).
\end{equation*}
From the last two equalities it follows that
\begin{equation*}
W_k^{(2s)}(k) -V_k^{(2s)}(k) =k f_{2s+9}(k^2), \quad V_k^{(2s+1)}(k) + W_k^{(2s+1)}(k) =k f_{2s+10}(k^2),
\end{equation*}
which proves (\ref{eq3.24}).
\end{proof}
Denote $w_s:=\lim_{k\rightarrow+0}\tfrac{d^sW(k)}{dk^s}$ and let
\begin{equation*}
W_{m_0-1}(k) = w_0 + w_1k + \dfrac{w_2}{2!}k^2 + \dots + \dfrac{w_{m_0-1}}{(m_0-1)!}k^{m_0-1}
\end{equation*}
be the Taylor polynomial for $W(k)$. From the previous lemma it follows that
\begin{equation*}
V(k) = W_{m_0-1}(-k) + o(k^{m_0-1}).
\end{equation*}
Since $k=0$ is an interior point of the spectrum, implying that $W(0) \neq 0$, and since by definition $R(k) = -\tfrac{V(k)}{W(k)}$, we see that
\begin{equation*}
R^{-1}(k) = R_{m_0-1}(-k) + o(k^{m_0-1}),
\end{equation*}
where $R_{m_0-1}(k)$ is a Taylor polynomial for $R(k)$ of degree $m_0-1$. Using  \eqref{eq3.23} we get
\begin{equation*}
P_1(k) = \dfrac{\I}{2} \left( R_{m_0-1}(k) - \ol{R_{m_0-1}(-k)} \right) + o(k^{m_0-1}),
\end{equation*}
which proves \eqref{eq3.22}.

The next step consists of verifying how often $F_R$ and $F_T$ can be integrated by parts.
If the initial data belongs to the class $\eqref{inithor}$, the first integral of $F_T$ can be integrated by parts $m_0-1$ times,
\begin{equation*}
F_T(x, t) = A(x, t) + O(e^{-\frac{c}{2}x}) + \dfrac{1}{x^{s-1}}\int_c^0 \dfrac{\pa^{s-1} (P\phi)}{\pa h^{s-1}}e^{-hx}dh,
\end{equation*}
where $A(x, t)$ corresponds to integrands at $h=0$. We have proven already that it cancels out with the integrand terms of $F_R$ at  $k=0$. The function $P$ has only $s-1$ derivatives with respect to $h$ and this procedure stops. The  integral can be differentiated by $x$ any number of times.

Each time we integrate $F_R$ by parts, there appears a new multiplier $3\I k^2t$ in front of $R(k)$, since we 
differentiate $\psi(-\I k)=e^{\I k^3t}$. As the integrand function should remain summable in $L_2(\R_+)$, 
we require $k^{2s-1}R(k) \in L_2(\R_+)$.
If the initial data has enough derivatives, $F_R$ can be integrated $s-1$ times too. Since we are searching for 
the classical solution of KdV, it has to have at least three derivatives. Therefore we require 
$k^{2s+2}R(k) \in L_2(\R_+)$. In summary,
\begin{lemma}\label{got}
Let 
\[\int_0^{+\infty} (1 +|x|^{s})\left(\left|\frac{d^n}{d x^n}q(x)\right| +\left|\frac{d^n}{d x^n}(q(-x)+c^2)\right|\right)dx<\infty,
\quad  0\leq n\leq l, \ s\geq 3. 
\] 
Then for any $m \leq s-1$ and $l \geq 2m+3$, the function $F(x, t)$  can be represented as
\begin{equation*}
F(x, t) = \dfrac{D(x, t)}{x^m} + D_1(x,t),\quad x\rightarrow +\infty, 
\end{equation*}
where $\frac{\pa^j D(x, t)}{\pa x^j} \in L_2(a,+\infty)$,  $0 \leq j \leq l -2m+1$, and 
$\frac{\pa^i D_1(x, t)}{\pa x^i}=O(e^{-\frac{c}{2}x})$ for $i \geq 0$, uniformly for $t \in [-\mathcal{T}, \mathcal{T}]$.
\end{lemma}

\subsection{Asymptotic behavior of the left kernel $F_1(x, t)$ in \eqref{defF1}}\label{subsection2}

We discard terms corresponding to the discrete spectra because they are exponentially small. The right Jost solution and hence the Wronskians in the expressions for $T, T_1$ and $R, R_1$ are functions of the local parameter $\sqrt{k_1 \mp c}$. Thus, if we want to integrate by parts, we need to differentiate the left reflection coefficient by $k_1$, which leads to singularities of type
\begin{equation*}
\dfrac{\pa^s \mathcal{R}(k_1)}{\pa k_1^s} = O\left(\left(k_1\mp c\right)^{-\frac{2s-1}{2}}\right), \quad s\geq 1 \text{ and } k_1 \rightarrow \pm c,
\end{equation*}
where $\mathcal{R}(k_1)=R_1(\la, t)$. Let $0 < \varepsilon < \frac{c}{8}$ and introduce 
\begin{equation*}
\mathcal B_{\pm}(k_1):= \mathcal B\left(\dfrac{k_1 \mp c}{\varepsilon}\right), \quad \text{ where }\mathcal  B(\xi) =
\begin{cases}
(1 - \xi^{2m_0})^{2m_0}, &\text{if } |\xi| \leq 1,\\
0, &\text{if } |\xi| \geq 1.
\end{cases}
\end{equation*}
Obviously $\frac{d^s\mathcal B_{\pm}}{dk_1^s} (\pm c+ \varepsilon) = \frac{d^s\mathcal B_{\pm}}{dk_1^s} ( \pm c-\varepsilon)= 0$, and $\frac{d^s\mathcal B_{\pm}}{dk_1^s} (\pm c) = 0$  for $0\leq s \leq 2m_0-1$.

We rewrite the integral containing $R_1$ in  \eqref{defF1} as 
\[ %label{eq3.25}
\begin{aligned}
& \int_{\R} R_1(\la)e^{-8ik_1^3t+12ic^2k_1t}e^{-ik_1x}dk_1 = \int_{\R} \left(1 - \mathcal B_+(k_1) - \mathcal B_-(k_1) \right) \mathcal{R}(k_1)e^{-ik_1x}dk_1  \\
& \qquad + \int_{-c-\varepsilon}^{-c+\varepsilon} \mathcal B_-(k_1) \mathcal{R}(k_1)e^{-ik_1x}dk_1 + \int_{c-\varepsilon}^{c+\varepsilon } \mathcal B_+(k_1) \mathcal{R}(k_1)e^{-ik_1x}dk_1 \\
&\quad =: I_1(x,t) + I_2^-(x, t) + I_2^+(x,t).
\end{aligned}
\]

The function $1 - \mathcal B_+(k_1) - \mathcal B_-(k_1)$ has zeros of degree $2m_0-1$ at $k_1=c$ and $k_1 = -c$ and is infinitely differentiable. Since the behavior of $\frac{\pa^m}{\pa k_1^m} (\mathcal{R}(1 - \mathcal B_+ - \mathcal B_-))$ as $k_1 \rightarrow \infty$ is the same as that of $\frac{\pa^m \mathcal{R}(k_1)}{\pa k_1^m}$, $k_1 \rightarrow \infty$, the integral $I_1(x, t)$ 
can be integrated by parts similarly to our previous considerations. 
To evaluate $I_2^{\pm}$, we  focus on $I_2^+$, the evaluation of $I_2^-$ is done in the same way. We start with
the Taylor series for $\mathcal{R}(k_1)$ on the interval $[-\varepsilon+c, \varepsilon+c]$. Let $z^2 = k_1 - c$, then 
\begin{equation*}
\mathcal{R}(c + z^2) = a_0(t )+ a_1(t)z + a_2(t)z^3 + \dots + a_{m_0-1}(t)z^{m_0-1} + \beta(z, t),
\end{equation*}
and $\beta(z, t) = O(z^{m_0})$ as $z \rightarrow 0$. Thus, $\mathcal{R}$ has at least $[\frac{m_0}{2}]$ derivatives with respect to $k_1$ at the point $k_1 = c$. Since $\mathcal B_+$ and its derivatives disappear at $k_1 = -\varepsilon+c$ and $k_1 = \varepsilon + c$, integration by parts gives
\begin{equation*}
I_2^+(x, t) = e^{-\I cx} \sum_{j=0}^{m_0-1}a_j(t) \int_{-\varepsilon}^{\varepsilon} \zeta^{\frac{j}{2}} \mathcal B\left(\dfrac{\zeta}{\varepsilon}\right)e^{-\I x\zeta}d\zeta + \dfrac{h_+(x, t)}{x^{[\frac{m_0}{2}]}},
\end{equation*}
where $h_+(x, t) \in L_2(\R_-)$, it is infinitely differentiable with respect to $x$, and all of the derivatives are also in $L_2(\R_-)$.

Let $\xi = \frac{\zeta}{\varepsilon}$. Using $1 - \xi^{2m} = (1 - \xi^2)(1 +\xi^2 + \xi^{4} + \dots + \xi^{2m-2})$ we get
\begin{equation*}
\begin{aligned}
\left(I_2^+(x, t)  - \dfrac{h_+(x, t)}{x^{[\frac{m_0}{2}]}}\right)\E^{\I c x} =& \sum_{j=0}^{[\frac{m_0}{2}]} \tilde{a}_{2j}(\varepsilon, t) \int_{-1}^1 \xi^{j}\mathcal B(\xi) e^{-\I x \xi \varepsilon} d\xi 
\\
& + \sum_{j=1}^{2m_0-2+[\frac{m_0}{2}]} \tilde{a}_{2j-1}(\varepsilon, t) 
\int_{-1}^1 \xi^{\frac{2j-1}{2}} (1-\xi^2)^{2m_0} e^{-\I x \xi \varepsilon} d\xi.
\end{aligned}
\end{equation*}
Each term of the first sum can be integrated by parts at least $2m_0-1$ times, moreover, all  integrands are equal to zero. All the summands of the second sum, except for the first one,  can be integrated by parts until we get the integral 
\begin{equation} \label{eq3.26}
\begin{aligned}
&J\left(x, 2 m_{0}\right) = \int_{-1}^{1} \xi^{\frac{1}{2}}\left(1-\xi^{2}\right)^{2 m_{0}} \mathrm{e}^{-\mathrm{i} \xi x \varepsilon} d \xi \\
&\quad = \frac{1}{2}\mathcal B(\nu, \rho)\ _1F_{2}\bigg(\nu, \frac{1}{2}, \nu+\rho, \frac{\mu^{2}}{4}\bigg)
+\frac{\mu}{2} \mathcal B\bigg(\nu+\frac{1}{2}, \rho\bigg)\ _1F_{2}\bigg(\nu+\frac{1}{2}, \frac{3}{2}, \nu+\rho+\frac{1}{2}, \frac{\mu^{2}}{4}\bigg) \\
&\qquad+\frac{\I}{2} \mathcal B(\nu, \rho)\ _1F_{2}\bigg(\nu, \frac{1}{2}, \nu+\rho, \frac{\mu^{2}}{4}\bigg)
-\frac{\mu \mathrm{i}}{2} \mathcal B\bigg(\nu+\frac{1}{2}, \rho\bigg)\ _1F_{2}\bigg(\nu+\frac{1}{2}, \frac{3}{2}, \nu+\rho+\frac{1}{2}, \frac{\mu^{2}}{4}\bigg),
\end{aligned}
\end{equation}
where $\mu = \I|x\varepsilon|$, $2\nu-1 = \frac{1}{2}$ and $\rho = 2m_0+1$. Here
\begin{equation*}
\mathcal B(\nu, \rho)=\frac{\Gamma(\nu) \Gamma(\rho)}{\Gamma(\nu+\rho)}
\end{equation*}
is the beta function, $\Gamma(\gamma)$ is the gamma function and $_1F_{2}(a, b, c, -z)$ is the generalized hypergeometric function. Using its asymptotic behavior for large  $|z|$  we get (cf.\ \cite[Equ.\ 16.11.8]{Olver})
\begin{equation} \label{eq3.28}
_1F_{2}(a, b, c,-z)=\frac{\Gamma(b) \Gamma(c)}{\Gamma(a)}\left(H_{1,2}(z)+E_{1,2}\left(z \mathrm{e}^{-\pi \mathrm{i}}\right)+E_{1,2}\left(z \mathrm{e}^{\pi \mathrm{i}}\right)\right)
\end{equation}
where
\begin{equation} \label{eq3.29}
\begin{aligned}
H_{1,2}(z)&=\sum_{k=0}^{c-a-1} \frac{\Gamma(a+k)}{\Gamma(b-a-k) \Gamma(c-a-k)} z^{-a-k} \frac{(-1)^{k}}{k !},
\\
E_{1,2}(z)&=\frac{1}{\sqrt{2 \pi}} 2^{-\alpha-(1 / 2)} \mathrm{e}^{2 z^{1 / 2}} \sum_{n=0}^{\infty} c_{n}\left(2 z^{1 / 2}\right)^{\alpha-n}, \quad \alpha=a-b-c+\frac{1}{2},
\end{aligned}
\end{equation}
and $c_n$ are constants. In \eqref{eq3.26} we substitute $a=\nu$, $b=\frac{1}{2}$, $c=\nu+\rho$ or $a=\nu+\frac{1}{2}$,  $b=\frac{3}{2}$, $c=\nu+\rho+\frac{1}{2}$ and also $\frac{\mu^{2}}{4}=:-z$, $\frac{-\mu \mathrm{i}}{2}=\sqrt{z} \in \R_{+}$. Then $\left(z \mathrm{e}^{\pm \I}\right)^{1 / 2} \in \mathrm{i} \R$, that is, the module of the last two terms in \eqref{eq3.28} containing the exponent is equal to 1. On the other hand,  in (\ref{eq3.29}) either $\alpha = -\rho = -2m_0 -1$ or $\alpha = -\rho - 1$. In any case, the terms containing $E_{1,2}$ can be evaluated as $ \frac{1}{(\varepsilon x)^{m_{0}+1 / 2}}($const.$+o(1))$. Thus, we can discard these terms. Combining the first  and the last summand of (\ref{eq3.26}) (the two other summands give the same function up to the multiplier $\I$) yields
\begin{equation*}
\begin{aligned}
&\frac{1}{2} \mathcal B(\nu, \rho)\ _1F_{2}\bigg(\nu, \frac{1}{2}, \nu+\rho,-z\bigg) \\
&\quad +z^{1 / 2}\mathcal B\bigg(\nu+\frac{1}{2}, \rho\bigg)\ _1F_{2}\bigg(\nu+\frac{1}{2}, \frac{3}{2}, \nu+\rho+\frac{1}{2},-z\bigg)=:
A_{1}+A_{2},
\\
&_1F_{2}\bigg(\nu, \frac{1}{2}, \nu+\rho,-z\bigg) \sim \frac{\Gamma\left(\frac{1}{2}\right) \Gamma(\nu+\rho)}{\Gamma(\nu)} \sum_{k=0}^{\rho-1} \frac{\Gamma(\nu+k)}{\Gamma\left(\frac{1}{2}-\nu-k\right) \Gamma(\rho-k)} z^{-\nu-k} \frac{(-1)^{k}}{k !},
\end{aligned}
\end{equation*}
that is,
\[
A_{1}=\frac{1}{2} \frac{\Gamma(\nu) \Gamma(\rho)}{\Gamma(\nu+\rho)} \frac{\Gamma\left(\frac{1}{2}\right) \Gamma(\nu+\rho)}{\Gamma(\nu)} \sum_{k=0}^{\rho-1} \frac{\Gamma(\nu+k)}{\Gamma\left(\frac{1}{2}-\nu-k\right) \Gamma(\rho-k)} z^{-\nu-k} \frac{(-1)^{k}}{k !}.
\]
Similarly,
\begin{equation*}
\begin{aligned}
A_{2} &=z^{1 / 2} \frac{\Gamma\left(\nu+\frac{1}{2}\right) \Gamma(\rho)}{\Gamma\left(\nu+\frac{1}{2}+\rho\right)} \frac{\Gamma\left(\frac{3}{2}\right) \Gamma\left(\nu+\frac{1}{2}+\rho\right)}{\Gamma\left(\nu+\frac{1}{2}\right)}
\\
& \qquad \times \sum_{k=0}^{\rho-1} \frac{\Gamma\left(\nu+\frac{1}{2}+k\right)}{\Gamma\left(\frac{3}{2}-\nu-\frac{1}{2}-k\right) \Gamma(\rho-k)} z^{-\nu-\frac{1}{2}-k} \frac{(-1)^{k}}{k !}.
\end{aligned}
\end{equation*}
Summing up these equalities and taking into account that $\nu = \frac{3}{4}$, we get
\begin{equation*}
\begin{aligned}
&A_{1} +A_{2}=\sum_{k=0}^{\rho-1} \frac{(-1)^{k}}{k !} z^{-\nu-k} \frac{\Gamma(\rho) \Gamma\left(\frac{3}{2}\right)}{\Gamma(\rho-k)}\left(\frac{\Gamma(\nu+k)}{\Gamma\left(\frac{1}{2}-\nu-k\right)}+\frac{\Gamma\left(\nu+\frac{1}{2}+k\right)}{\Gamma(1-\nu-k)}\right)+
\\
&+O\big(z^{-\nu-\rho+1}\big) \sum_{k=0}^{\rho-1} \frac{(-1)^{k}}{k !} z^{-\nu-k} \frac{\Gamma(\rho) \Gamma\left(\frac{3}{2}\right)}{\Gamma(\rho-k)}\left(\frac{\pi}{\sin \left(\frac{3 \pi}{4}+\pi k\right)}+\frac{\pi}{\sin \left(\frac{3 \pi}{4}+\frac{\pi}{2}+\pi k\right)}\right)
\\
&+O\big(z^{-\nu-\rho+1}\big)=O\big(x^{-2 m_{0}-1}\big).
\end{aligned}
\end{equation*}
Since $E_{1,2} (ze^{\pm i \pi})$ have the same order, we conclude that 
$J(x, 2m_0) = O(x^{-m_0 - \frac{1}{2}})$. Also
\begin{equation*}
\frac{d}{d x} J(x, 2 m_{0})=\frac{1}{\mathrm{i} \varepsilon x}\left(J(x, 2 m_{0})-4 m_{0} \int_{-1}^{1} \xi^{3 / 2}\left(1-\xi^{2}\right)^{2 m_{0}-1} \mathrm{e}^{-\mathrm{i} \xi(x \varepsilon)} d \xi\right),
\end{equation*}
and the last integral is evaluated as $O\big(x^{-2} J(x, 2 m_{0}-2)\big)$ (plus terms of higher order). Therefore, derivatives of $J(x, 2m_0)$ decrease even faster. Hence the main contribution to $I_2^{\pm}(x, t)$ comes from the term 
$h_{\pm}(x, t)x^{-[\frac{m_0}{2}]}$. That  is why we can integrate $I_1(x, t)$ by parts exactly $[\frac{m_0}{2}]$ times.
The behavior of the integrand as $k_1\to \pm\infty$ after the last integration is evaluated as $k_{1}^{2\left[\frac{m_{0}}{2}\right]-n_{0}-1} f\left(k_{1}\right) \mathrm{e}^{-\mathrm{i} k_{1} x}$, where $f \in L_2(\pm\infty)$. Every differentiation of $I_1(x, t)$ with respect to $x$ will increases the degree $2\left[\frac{m_{0}}{2}\right]-n_{0}-1 $ by 1. To obtain a classical solution we need to differentiate at least 4 times. Thus, we have to require $m_0 - n_0 - 1 \leq -4$, or $n_0\geq m_0 +3$. The following lemma sums up these considerations.
\begin{lemma}
Let $q(x)$ satisfy \eqref{inithor} with $n_0\geq m_0 +3$. Then $F_1(x, t)$ admits the representation as $x\rightarrow -\infty$
\[
F_{1}(x, t)=\frac{H_{1}(x, t)}{ x^{[\frac{m_0}{2}]}}, \quad \frac{\partial^{j} H_{1}(x, t)}{\partial x^{j}} \in L_{2}(-\infty, -a),
\quad j=0, \ldots, n_{0}-m_{0}+1,
\] 
with $a\gg 1$.
\end{lemma}

From Lemma \ref{got} we obtain
\begin{lemma}
Let $q(x)$ satisfy \eqref{inithor} with $n_0\geq m_0 +3$. Then $F(x, t)$ admits the representation
as $x\rightarrow +\infty$
\[
F(x, t)=\frac{H(x, t)}{ x^{[\frac{m_0}{2}]}}, \quad \frac{\partial^{j} H(x, t)}{\partial x^{j}} \in L_{2}(a, +\infty), 
\quad j=0, \ldots, n_{0}-m_{0}+1,
\]
with $a\gg 1$.
\end{lemma}
Since $\frac{1}{x} \frac{\partial^{j} H(x, t)}{\partial x^{j}} \in L_{1}(a,+\infty) $ and $ \frac{1}{x} \frac{\partial^{j} H_{1}(x, t)}{\partial x^{j}} \in L_{1}(-a, -\infty)$, both $F_x^{(j)}(x, t)$ and $F_{1, x}^{(j)}(x, t)$ satisfy condition {\bf IV} with $m_0(t) = [\frac{m_0}{2}]-1$ and $n_0(t) = n_0 - m_0$. Naturally, we assume that $m_0(t)\geq 1$, i.e., $m_0\geq 4$. This proves Theorem~\ref{thoern}. 

\noprint{

It remains to show the case $m_0 =3$ and $n_0\geq 6$ of Theorem \ref{thoern}.
Properties {\bf I.}--{\bf III.} can be verified as above. Moreover,
\begin{equation*}
\begin{aligned}
&F_{x}^{(j)}(\cdot, t) \in L^{1}\left(\mathbb{R}_{+}\right) \cap
L_{loc}^{1}(\mathbb{R}), \quad F_{1, x}^{(j)}(\cdot, t) \in L^{1}\left(\mathbb{R}_{-}\right) \cap L_{loc}^{1}(\mathbb{R}), \quad j=0, \dots, 4,
\\
&\frac{\partial}{\partial t} F(\cdot, t) \in L^{1}\left(\mathbb{R}_{+}\right) \cap L_{\operatorname{loc}}^{1}(\mathbb{R}), \quad \frac{\partial}{\partial t} F_{1}(\cdot, t) \in L^{1}\left(\mathbb{R}_{-}\right) \cap L_{\mathrm{loc}}^{1}(\mathbb{R}).
\end{aligned}
\end{equation*}
We need to find a monotonous majorant for $|F(x, t)|$ and $|F_1(x, t)|$. For $F(x, t)$ we have shown that $F(x, t) \in L^{1}\left(\mathbb{R}_{+}\right) \cap L_{\operatorname{loc}}^{1}(\mathbb{R})$ has $4$ derivatives. Having $6$ derivatives and $3$ moments of the initial data allows us to integrate by parts twice. Differentiating by $x$ yields
\begin{equation*}
\frac{\partial}{\partial x} F(x, t)=-\frac{1}{2 \pi x^{2}}\left(\mathrm{i} \int_{-\infty}^{\infty} k \frac{\partial^{2}(R \psi)}{\partial k^{2}} e^{\mathrm{i} k x} d k+2 \int_{c / 2}^{0} h \frac{\partial^{2}(P \psi)}{\partial h^{2}} e^{-h x} d h\right)+O\bigg(\frac{1}{x^{3}}\bigg),
\end{equation*}
where the term $O(x^{-3})$ contains the summands corresponding to the discrete spectrum. Since $k \frac{\partial^{2}(R \psi)}{\partial k^{2}} = O\left(k^{-2}\right)$ as $k\rightarrow \infty$ and the 
second derivatives of $P$ and $R$ are continuous (recall that the initial data has  $3$ finite moments), 
we have that $x \frac{\partial}{\partial x}F(x, t) \in L^1(\R_+)$. Similarly, $\frac{\pa}{\pa x}I_1(x, t)$ has the first finite  moment. To estimate $I_2^\pm(x,t)$ in \eqref{eq3.25} we will use the following
\begin{lemma} \label{lem3.6} (\cite{Stein}, corollary on p. 334 )
Let $g$ be a smooth real function defined on $(a, b)$ and $|g''(k)|\geq 1$ for all $k \in (a, b)$. Then for $y>0$
\begin{equation*}
\left|\int_{a}^{b} \mathrm{e}^{\mathrm{i} y g(k)} \Psi(k) d k\right| \leq C \frac{1}{\sqrt{y}}\left[|\Psi(b)|+\int_{a}^{b}\left|\Psi^{\prime}(k)\right| d k\right]
\end{equation*}
and the constant $C$ does not depend on $g$ or $y$.
\end{lemma}
We consider $I_2^+(x, t)$ from (\ref{eq3.25}) and denote
\begin{equation*}
\widetilde{\Psi}(k_{1}):=R_{1}(\lambda) B_{+}(k_{1}), \quad \widetilde{g}(k_{1})=k_{1}+v(3 c^{2} k_{1}-2 k_{1}^{3}), \quad v:=\frac{4 t}{x}.
\end{equation*}
\todo[inline, color=blue!10 ]{It should be $\widetilde{g}(k_{1})=k_{1}+v(- 3 c^{2} k_{1} + 2 k_{1}^{3})$.\\

}
Note that $v$ is small for $t \leq \mathcal{T}$ and $x \rightarrow -\infty$. We differentiate $\hat{g}(k)=\widetilde{g}(k_{1})$ twice with respect to $k = \sqrt{k_1^2 -c^2}$ and evaluate it near $k=0$,
\begin{equation*}
\frac{d \hat{g}}{d k}=\frac{k}{k_{1}}\big(1+v\big(3 c^{2} k_{1}-6 k_{1}^{3}\big)\big), \quad \frac{d^{2} \hat{g}}{d k^{2}}=\frac{1+v\big(3 c^{2} k_{1}-6 k_{1}^{3}\big)}{k_{1}}+O(k).
\end{equation*}
Since in $I_2^+(x, t)$ we have $|k|< C \sqrt{\varepsilon}$ and $k_{1} \geq \pm c-\varepsilon$, $v$ is small, there exists a constant $0<\mathcal{K}<\frac{4}{c}$, such that $\frac{d^{2} \hat{g}}{d k^{2}}>\mathcal{K}>0$ inside the area of integration. Rewriting $I_2^+(x, t)$ yields
\begin{equation*}
\begin{aligned}
I_{2}^{+}(x, t)&=\int_{-\varepsilon+c}^{\varepsilon+c} \widetilde{\Psi}(k_{1}) \mathrm{e}^{-\I x \widetilde{g}(k_{1})} d k_{1}\\
&=\int_{\left[\mathrm{i} \kappa_{\varepsilon}, 0\right] \cup\left[0, \kappa_{\varepsilon}\right]} \frac{\widetilde{\Psi}\left(k_{1}\right)}{1+v(3 c^{2} k_{1}-6 k_{1}^{3})} \hat{g}^{\prime}(k) \mathrm{e}^{-\I x \hat{g}(k)} d k
\end{aligned}
\end{equation*}
\todo[inline, color=blue!10 ]{It should be $\kappa_{\varepsilon}$, not $\I \kappa_{\varepsilon}$, in the boundary of the integral.
\\\ NO, it should be $\i\kappa_\epsilon$, because $k$ in pure imaginary when $c-\epsilon<k_1<c$.}
where $\kappa_{\varepsilon} = \sqrt{2c\varepsilon + \varepsilon^2}$. We can integrate this integral by parts. The properties of $B_+$ ensure the disappearance of terms outside the integral,
\be \label{I_2 final}
I_{2}^{+}(x, t)=\frac{1}{\mathrm{i} x} \int_{\left[\I \kappa_{\varepsilon}, 0\right] \cup\left[0, \kappa_{\varepsilon}\right]} \Psi(k) \mathrm{e}^{\mathrm{i} y g(k)} d k,
\ee
where
\begin{equation*}
\Psi(k):=\frac{d}{d k} \frac{R_{1}(\lambda) B_{+}\left(k_{1}\right)}{1+v(3 c^{2} k_{1}-6 k_{1}^{3})}, \quad y=-\mathcal{K} x, \quad g(k)=\mathcal{K}^{-1} \hat{g}(k).
\end{equation*}
We know that if $m \geq 3$, $R_1(\la)$ has two continuous derivatives by $k$ in the neighborhood of $k_1 = c$.
The other functions included in $\Psi(k)$ have an infinite number of derivatives, and they are bounded.
Changing to the variable $\xi = -\mathrm{i} k$ in \eqref{I_2 final} and using Lemma~\ref{lem3.6} gives 
us the required estimate $|I_{2}^{+}(x, t)| \leq const \ |x|^{-3 / 2}$. Similarly, $|I_{2}^{-}(x, t)| 
\leq const \ |x|^{-3 / 2}$. 
\todo[color=blue!10 ]{$\mathcal{L}_3^6 (0, -c^2)$ is not defined. }
Thereby we proved that if $q(x) \in \mathcal{L}_3^6 (0, -c^2)$, 
there exists a unique solution $q(x, t)$ of the   Cauchy problem for KdV with initial data $q(x)$ such that 
\begin{equation*}
\int_{\mathbb{R}_{+}}\left(|q(x, t)|+|q(-x, t)+c^{2}|\right) d x<\infty, \quad
\frac{\partial q(\cdot, t)}{d t}, \frac{\partial^{j} q(\cdot, t)}{d x^{j}} \in L^{1}(\mathbb{R}), \quad j=1,2,3.
\end{equation*}
This finishes the proof of Theorem~\ref{thoern}.
}

%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Proof of Theorem \ref{maintheor1}.}\label{sec4}
We represent the right Marchenko equation \eqref{March1} in the form (compare \cite{DT})
\[ B(x,y,t) + \hat F(x+y, t) +\int_0^\infty B(x,s,t)\hat F(x+y+s,t)ds=0,
\]
where
\[
B(x,y,t)=2K(x, x+2y,t),\quad \hat F(x,t)= 2F(2x,t).
\]
The kernel $\hat F(x,t)$ consists of three summands,
\begin{align*} \hat F_T(x,t)&=\dfrac{2}{\pi}\int_c^0 P(h) \E^{8 h^3 t -2 h x} dh, \quad
\hat F_d(x,t)= 2\sum_{j=1}^N\gamma_j^2\E^{-2\kappa_j x+8\kappa_j^3 t}, \\
\hat F_R(x,t)&=\dfrac{2}{\pi}\re\int_0^{+\infty} R(k)\E^{8\I k^3 t + 2 \I k x}dk, 
\end{align*}
where $k=\I h$ and $P(h)$ is defined by \eqref{eq3.19}. 
The following result is well known.
\begin{lemma}[\cite{Zakh}] \label{zak}  Let $\delta_{ij}$ be the Kronecker symbol and let $A(x,t)$ be a 
$N\times N$ matrix with elements
\[
A_{ij}(x,t)=\delta_{ij} +\frac{\gamma_j^2\E^{8\kappa_j^3 t}}{\kappa_i +\kappa_j}\E^{-(\kappa_j + \kappa_j)x}.
\]
Let $A^{(j)}(x,t)$ be the matrix obtained from $A(x,t)$ by replacing the $j$-th column of $A$ with 
\[
\begin{pmatrix}
-\gamma_1^2\E^{8\kappa_1^3 t -\kappa_1 x} \\
%\vdots \\
 %-\gamma_i^2\E^{8\kappa_i^3 t -\kappa_i x} \\
 \vdots \\
 -\gamma_N^2\E^{8\kappa_N^3 t -\kappa_N x}
\end{pmatrix}
\]
Then the Marchenko equation with kernel $\hat F_d(x,t)$,
\be\label{nuzh}
B_d(x,y,t) + \hat F_d(x+y, t) +\int_0^\infty B_d(x,s,t)\hat F_d(x+y+s)ds=0,
\ee
has a unique solution $B_d(x,y,t)$ such that
\[
B_d(x,0,t)=\frac{1}{\det A(x,t)}\sum_{j=1}^N \det A^{(j)}(x,t)\E^{-\kappa_j x}.
\]
The reflectionless, fast decaying solution $u(x,t)=-\frac{\pa B_d(x,0,t)}{\pa x}$ of the KdV equation associated with \eqref{nuzh} can be expressed as
\[
u(x,t)=-2\frac{\pa^2}{\pa x^2}\log\det A(x,t).
\]
In the domain $x>\varepsilon t$, the following asymptotic is valid as $t\to +\infty$ with $C>0$,
\be\label{solit} 
u(x,t)=-\sum_{j=1}^N\frac{2\kappa_j^2}{\cosh^2\left(\kappa_j x - 4\kappa_j^3 t -\frac{1}{2}\log\frac{\gamma_j^2}{2\kappa_j}-\sum_{i=j+1}^N\log\frac{\kappa_j - \kappa_i}{\kappa_i + \kappa_j}\right)} + O(\E^{-C t}).
\ee
\end{lemma}
Our aim is to prove that the main contribution to $\hat F(x,t)$ in the region 
\be\label{mathd}\mathcal O:=\left\{ (x,t): t\geq \mathcal T,\  x\geq 4c^2 t + \log t^{\frac{m_0-3/2 -\varepsilon}{c}}\right\},
\ee  
stems from $\hat F_d(x,t)$.  To this end we have to estimate $\hat F_T(x+y,t) +\hat F_R(x+y,t)$ for $(x,t)\in \mathcal O$, $y\geq 0$.
  Since $t$ is arbitrary large, we  cannot use 
integration by parts as in the previous section. Denote  
\[
\aligned x=4c^2 t& +\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t + \xi,\quad  r=2(\xi+y),\\
  S(r,t,h)&:=8 h^3 t - 8hc^2 t -h\frac{2m_0-3-2\varepsilon}{c}\log t - h r.
  \endaligned\]
Note that for $t \geq \mathcal T$, $r\geq 0$, and $0\leq h\leq \frac{c}{2}$,
\[
\aligned S_h(r,t,h):=\frac{\pa S(r,t,h)}{\pa h}&=24h^2 t   -8c^2 t- 2\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t - r<0, \\
\left|\frac{1}{S_h(r,t,h)}\right|&\leq \frac{1}{2c^2 t + r +1},
\endaligned
\] 
where $\mathcal T\geq\E^{\frac{c}{2m_0-3-2\varepsilon}}$.
 Split the integral in $\hat F_T(x+y,t)$ into 
two parts, 
\begin{align} \label{razbi}
F^{(1)}(x+y,t)&=\frac{2}{\pi}\int_c^{c/2} P(h) \E^{S(r,t,h)} dh, \\ \label{rzd}
F^{(2)}(x+y,t)&=\frac{2}{\pi}\int_{c/2}^0 P(h)\E^{S(r,t,h)}dh= \frac{2}{\pi}
\int_{c/2}^0 \frac{P(h)}{S_h(r,t,h)}d\E^{S(r,t,h)}.
\end{align}
Since
\[ S(r,t,h)\leq -\Big(m_0-\frac 3 2 -\varepsilon\Big)\log t -\frac{c}{2}r,\quad \mbox{for}\ h\in [c/2, c],
\]
then
\be\label{estkl}\aligned | F^{(1)}(x+y,t)|&\leq \frac{2}{\pi} \frac{1}{t^{m_0-\frac 3 2 -\varepsilon}}\E^{-\frac{c}{2} (\xi +y)}\int^c_{c/2}|P(h)|d h,\\
\left| \frac{\pa}{\pa x} F^{(1)}(x+y,t)\right|&\leq \frac{2}{\pi} \frac{1}{t^{m_0-\frac 3 2 -\varepsilon}}\E^{-\frac{c}{2} (\xi +y)}\int^c_{c/2}|h P(h)|d h.\endaligned
\ee 
On the other hand, we observe that $\psi(r,t,h):=\frac{1}{S_h(r,t,h)}$  satisfies \eqref{eq3.20}. An elementary analysis shows that 
\be\label{estpsi1} 
\left|\frac{\pa^n}{\pa h^n}\psi(r,t,h)\right|\leq C|\psi(r,t,h)| \leq 
C\begin{cases} \begin{array}{ll}\frac{1}{2c^2t + r +1}, & h\in [0, \frac{c}{2}],\\
\frac{1}{24 k^2 t + 8c^2 t + r + 1}, & k=\I h\in \R_+.\end{array}\end{cases}
\ee
The positive constant $C$ does not depend on $t$ and $r$. Let us simultaneously integrate both integrals in the sum 
\be\label{11}\aligned I(x+y, t):=& \frac{\pi}{2}\left(F^{(2)}(x+y,t) + \hat F_R(x+y,t)\right)\\
=&\int_{c/2}^0 P(h)\psi(r,t,h)d\E^{S(r,t,h)} +\re\int_0^\infty R(k)\psi(r,t,-\I k) d\E^{S(r,t,-\I k)} \endaligned\ee  by parts $m_0$-times, using each time the representation
\[\E^{S(r,t,h)}dh=\psi(r,t,h) d \E^{S(r,t,h)}, \quad -\I \E^{S(r,t,-\I k)}dk= \psi(r,t,-\I k) d \E^{S(r,t,-\I k)}.
\] 
Taking into account \eqref{eq3.20}, \eqref{eq3.22}, \eqref{estpsi1} we get 
\[ \aligned
I(x+y, t)&=\E^{S(r,t,c/2)} \mathcal P_1\Big(r,t,\frac{c}{2}\Big)
+ \int_c^{c/2} \mathcal P_2(r,t,h)\left(\psi(r,t,h)\right)^{m_0-1}dh \\ 
&\quad +\re\int_0^\infty \mathcal P_3(r,t,-\I k)
\big(\psi(r,t,-\I k)\big)^{m_0-1} dk,
\endaligned\]
where the functions $\mathcal P_i(r,t,h)$ are uniformly bounded with respect to all arguments as $t\geq\mathcal T$, $r\geq 0$, $h\in [0, \frac c 2]$ and $k\in \R_+$.
Using again estimate \eqref{estpsi1} and inequality  $S(r,t,\frac c 2)< -7c^3 t - \frac{ c }{2} r$ we obtain
\be\label{estm1}\aligned |I(x+y,t)|& \leq C  \Big(\E^{-7c^3t}\E^{-\frac{c r}{2}} + \int_{c/2}^c \frac{dh}{\left(2c^2 t + r + 1\right)^{m_0-1}} +\int_{\R_+}\frac{dk}{\left(24k^2 t + 8c^2 t + r +1\right)^{m_0-1}}\Big).
\endaligned \ee

For the next step recall Young's inequality: For all $u>0, v>0$, $p>1$, $q>1$, such that $\frac{1}{p}+\frac{1}{q}=1$, one has  
\[ uv\leq \frac{u^p}{p} +\frac{v^{q}}{q}.
\]
In the last integral in \eqref{estm1} we choose 
 $u^p=p\cdot (24 k^2 t +8c^2 t)$, $v^q=q \cdot(r+1)$. In the first integral $v$ is the same and $u^p=p\cdot 2c^2 t$. 
Then
\be\label{13}\aligned  |I(x+y,t)|&\leq C\left(\E^{-7c^3t}\E^{-\frac{c r}{2}} +\frac{c}{2(2pc^2)^{\frac{m_0-1}{p}}\cdot q^{\frac{m_0-1}{q}}}\frac{1}{ t^{\frac{m_0-1}{p}} (r+1)^{\frac{m_0-1}{q}}}\right.
\\ &+\left.\frac{1}{p^{\frac{m_0-1}{p}}\cdot q^{\frac{m_0-1}{q}}} \frac{1}{t^{\frac{m_0-1}{p}} (r+1)^{\frac{m_0-1}{q}}}\int_{\R_+}\frac{d k}{ (24 k^2 + 8 c^2)^{\frac{m_0-1}{p}}}\right).\endaligned
\ee
To achieve convergence in the last integral we have to require $m_0-1>\frac{p}{2}$. In later 
estimates we need the property $(r+1)^{-\frac{m_0-1}{q}}\in L_2(\R_+)$, which implies $m_0-1>\frac{q}{2}$. 
Set
\[ \frac{1}{q}=\frac{\frac 1 2 +\varepsilon}{m_0-1}, \quad \ \frac 1 p= 1 -\frac{1}{q},\quad \mbox{that is,} \quad  \frac{m_0-1}{p}=m_0-\frac 3 2 - \varepsilon,
\]
where $\varepsilon>0$ is arbitrary small. Combining this with \eqref{11} and \eqref{13} implies
\be\label{nui}\big|F^{(2)}(x+y,t) + \hat F_R(x+y,t)\big|\leq \ti C\frac{1}{t^{m_0-\frac 3 2 -\varepsilon}}\ \frac{1}{(\xi + y+1)^{\frac 1 2 +\varepsilon}},\ee
where $\ti C>0$ does not depend on $x,y,t$ in the region 
\[t\geq\mathcal T, \quad x\geq 4c^2 t +\frac{m_0-\frac 3 2-\varepsilon}{c}\log t,\quad y\geq 0.
\] 
Repeating almost literally the arguments above  for $\frac{\pa}{\pa x}(F^{(2)}(x+y,t) + \hat F_R(x+y,t))$ 
we get the estimate
\be\label{nui1}
\left|\frac{\pa}{\pa \xi}(F^{(2)}(x+y,t) + \hat F_R(x+y,t))\right|\leq \hat C\frac{1}{t^{m_0-\frac 3 2 -\varepsilon}}\ \frac{1}{(\xi + y+1)^{\frac 1 2 +\varepsilon}}.
\ee
Combining  \eqref{razbi}--\eqref{estkl} with \eqref{nui}, \eqref{nui1}  we finally obtain that
\begin{align}\label{mainest} 
& \big|\hat F_T(x+y,t) + \hat F_R(x+y,t)\big|\leq C\frac{1}{t^{m_0-\frac 3 2 -\varepsilon}}\ \frac{1}{(\xi + y+1)^{\frac 1 2 +\varepsilon}}, \\
\label{mainest1}
& \left|\frac{\pa}{\pa \xi}\big(\hat F_T(x+y,t) + \hat F_R(x+y,t)\big)\right|\leq C\frac{1}{t^{m_0-\frac 3 2 -\varepsilon}}\ \frac{1}{(\xi + y+1)^{\frac 1 2 +\varepsilon}}.
\end{align}
 
 Consider the space of functions $\varphi(y) \in L_2(\R_+) \cap C(\R_+)$ with the norm $\| \varphi \| = \| \varphi \|_{L_2} + \| \varphi\|_{C}$. In this space introduce the operators 
\[
\aligned \relax[ \mathcal F\varphi ](y) &=\int_{0}^{\infty} \hat F_d\Big(4c^2t +\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t+ \xi+y+s, t\Big) \varphi(s) d s, \\
[\mathcal G \varphi](y) &=\int_0^{\infty} G\Big(4c^2t +\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t + \xi+y+s, t\Big) \varphi(s) ds,
\endaligned 
\]
where  $G(x,t)=\hat F_T(x, t) +\hat F_R(x, t)$.
The Marchenko equation  \eqref{nuzh} can be represented as  
\begin{equation} \label{eq4.3}
\varphi+\F \varphi+\mathcal G \varphi=\omega+g,
\end{equation}
where 
\begin{align*}
\omega&=\omega(y)=-\hat F_d\Big(4c^2 t +\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t+\xi+y, t\Big),\\
 g&=g(y)=-\mathcal G\Big(4c^2 t+\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t+\xi+y, t\Big), \\
\varphi &=\varphi(y)=B\Big(4 c^{2} t+\frac{m_0-\frac 3 2 -\varepsilon}{c}\log t+\xi, y, t\Big).
\end{align*}
The operators $\F$, $\mathcal G$ and the functions $\omega$, $g$, $\varphi$ depend on $\xi$ and $t$ as parameters. By \eqref{mainest} and \eqref{mainest1} we have 
\begin{equation} \label{eq4.5}
\|G\|+\left\|G_{\xi}\right\| \leq C t^{-(m_0 - 3/2-\varepsilon)}, \quad \|g\|+\left\|g_{\xi}\right\| \leq  C t^{-(m_0 - 3/2-\varepsilon)}, 
\end{equation}
 where $G_{\xi} = \frac{dG}{d \xi}$, $g_{\xi} = \frac{dg}{d \xi}$, and the norm is taken in 
$L_2(0, \infty) \cap C[0, \infty)$. 

Lemma \ref{zak} has the following corollary (cf.\ \cite{Kh}).
\begin{corollary}\label{cor}
The operator $\id + \F$ in the spaces $L_2(\R_+)$ and $ L_2(\R_+) \cap C[\R_+)$ is invertible,  and the norms of the operators $\RR = (\id + \F)^{-1}$ and $\RR_{\xi} = \frac{d\RR}{d \xi}$ are bounded uniformly with respect to $t$ and $\xi \geq 0$.
\end{corollary}


 
From this corollary it follows that \eqref{eq4.3} can be rewritten as
\begin{equation*}
\varphi+\RR \mathcal G \varphi=\RR \omega+\RR g,
\end{equation*}
where the operator $\RR \mathcal G$ is small in the norm of $ L_2(\R_+) \cap C(\R_+)$ for large $t$.
Applying successive approximation we get
\[
\varphi=\RR  \omega+\RR  g-\sum_{n=1}^{\infty}(\RR \mathcal G)^{n}[\RR  \omega+\RR  g].
\]
Differentiating  by $\xi$ yields
\begin{equation}\label{eq4.9}
\begin{aligned}
\frac{d}{d \xi} \varphi&=\frac{d}{d \xi}(\RR \omega)+\RR _{\xi} g +\RR  g_{\xi} -\sum_{n=1}^{\infty} \sum_{i=0}^{n-1}(\RR  G)^{i}\left(\RR _{\xi} G+\RR  G_{\xi}\right)
(\RR  G)^{n-1-i}[\RR  \omega+\RR  g]
\\
&\quad -\sum_{n=1}^{\infty}(\RR  G)^{n}\left[\frac{d}{d \xi}(\RR  \omega)+\RR _{\xi} g+\RR  g_{\xi}\right].
\end{aligned}
\end{equation}
Taking into account estimates (\ref{eq4.5}) and Corollary \ref{cor} we obtain that the series on the r.h.s. of \eqref{eq4.9} converge in $L_2(\R_+) \cap C(\R_+)$ uniformly with respect to  $t$.
From (\ref{eq4.5}) and \eqref{eq4.9} it also follows that
\[
\frac{d}{d \xi} \varphi \Big|_{y=0}=\frac{d}{d \xi}\left(\RR \omega\right)\Big|_{y=0}+O\big(t^{-(m_0-3/2-\varepsilon)}\big).
\]
Therefore the main contribution in the asymptotics of the solution $q(x,t)$ of the initial value problem \eqref{kdv}, \eqref{inithor} is given by the solution of \eqref{nuzh}. Together with  \eqref{solit} this finishes the  proof of Theorem \ref{maintheor1}.

\section{Conclusion}

Using the classical IST we found minimal decaying and smoothness conditions on the steplike scattering data which guarantee the solvability of the associated RHP. This solvability requires that the KdV solution $q(x,t)$ tends to the background 
constants with finite first moment. In turn, this gives an analytically rigorous approach to using the nonlinear steepest 
descent method in the steplike case. The alternative approach to study solvability of RH problems is numerically, see for example Bilman and Trogdon (\cite{BT}).

In this work we also solved the more challenging technical problem of describing the largest class of steplike initial data for which the initial value problem solution $q(x,t)$ has a prescribed smoothness and a prescribed number of finite moments of perturbations. 

For possible applications, the most interesting result is that we justify the soliton asymptotics of steplike 
solutions in a larger region than previously known and for essentially larger classes of initial data. Such classes 
are connected with RH problems of low regularity, compare Lenells (\cite{Lenells}).





\end{document}
